PNNL-17439



D. R. Newcomer

March 2008



Pacific Northwest National Laboratory Operated by Battelle for the U.S. Department of Energy

> Prepared for the U.S. Department of Energy under Contract DE-AC05-76RL01830

DISCLAIMER

This report was prepared as an account of work sponsored by an agency of the United States Government. Neither the United States Government nor any agency thereof, nor Battelle Memorial Institute, nor any of their employees, makes **any warranty, express or implied, or assumes any legal liability or responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights**. Reference herein to any specific commercial product, process, or service by trade name, trademark, manufacturer, or otherwise does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof, or Battelle Memorial Institute. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or any agency thereof.

PACIFIC NORTHWEST NATIONAL LABORATORY operated by BATTELLE for the UNITED STATES DEPARTMENT OF ENERGY under Contract DE-AC05-76RL01830

Printed in the United States of America

Available to DOE and DOE contractors from the Office of Scientific and Technical Information, P.O. Box 62, Oak Ridge, TN 37831-0062; ph: (865) 576-8401 fax: (865) 576-5728 email: reports@adonis.osti.gov

Available to the public from the National Technical Information Service, U.S. Department of Commerce, 5285 Port Royal Rd., Springfield, VA 22161 ph: (800) 553-6847 fax: (703) 605-6900 email: orders@ntis.fedworld.gov online ordering: http://www.ntis.gov/ordering.htm



300 Area VOC Program Slug Test Characterization Results for Selected Test/Depth Intervals for Wells 399-2-5, 399-3-22, and 399-4-14

D. R. Newcomer

March 2008

Prepared for the U.S. Department of Energy under Contract DE-AC05-76RL01830

Pacific Northwest National Laboratory Richland, Washington 99352

Executive Summary

Multiple, stress-level slug tests were performed at selected test/depth intervals within wells 399-2-5, 399-3-22, and 399-4-14 as part of the 300 Area volatile organic compound characterization program at the Hanford Site in Washington State. The temporary test screen lengths were characterized as the boreholes were advanced to their final drill depths and before their completion as monitor-well facilities. Following well completion, slug tests were performed in the final, completed well-screen sections. The objectives of the slug tests were to provide the vertical distribution of hydraulic conductivity with depth at these locations and to support selection of the final well screen-depth interval for each of these monitor-well facilities. This characterization information is important for predicting/simulating contaminant migration (i.e., numerical flow/transport modeling) and designing proper monitor-well strategies within this area.

Test-analysis results obtained from the multiple, stress-level slug tests provide vertical distribution of hydraulic conductivity for hydrogeologic units generally within the upper, middle, and lower sections of the unconfined aquifer. Individual test/depth intervals were sited to provide hydraulic-property information for the highly permeable Hanford formation (Unit 1) within the upper part of the unconfined aquifer and the underlying, less permeable Ringold Formation (Unit 5) within the middle and lower sections of the unconfined aquifer. Eight out of 10 discrete-depth intervals were tested successfully during borehole advancement, and one test/depth interval was tested after the wells were completed as monitor-well facilities. Two of the temporary test screen lengths could not be tested during borehole advancement in-filling that occurred inside the temporary well-screen section.

No quantitative analysis for slug tests conducted within the three Hanford formation (Unit 1) test intervals was realized because of test-system limitations. Limiting qualitative analysis results, however, provide a lower, bounding hydraulic conductivity estimate range of \geq 300 to \geq 400 m/day for these Hanford formation tests. These hydraulic conductivity estimates were derived for test-interval sections that ranged from only 0.5 to 1.1 m in length. These lower bounding Hanford formation test values are comparable to the general range of lower bounding values (i.e., >100 to >2,000 m/day) for 300-Area test characterizations recently cited in Williams et al. (2007) and to the estimate of 568 m/day for one previous 300-Area volatile organic compound characterization test/depth interval (Spane 2007).

Analysis of the slug-tests conducted within six test/depth intervals within the Ringold Formation (Unit 5) indicates average hydraulic conductivity estimates ranging from ≤ 0.01 to 2.48 m/day. Hydraulic conductivity estimates for the Ringold Formation (Unit 5) were derived for test-interval sections that ranged from 0.6 to 2.9 m in length. These average hydraulic conductivity values are comparable to the lower range of 0.04 to 41.2 m/day, with a geometric mean of 2.38 m/day, for 16 other Ringold Formation test/depth intervals recently obtained for test-characterization boreholes in the 300 Area (Williams et al. 2007; Spane 2007).

Acknowledgments

Those acknowledged include field support provided by Rob Mackley of Pacific Northwest National Laboratory, Jake Horner of Gram, Inc., and field-testing personnel and test-equipment support provided by Blue Star Enterprises Northwest Drilling Company. Frank Spane provided technical peer review comments, Dave Lanigan provided graphics support, and Wayne Cosby provided editorial comments.

Contents

Sum	nmary	⁷		iii
Ack	nowl	edgme	nts	v
1.0	Intro	oductio	n	1.1
2.0	Hyd	rologie	e Test System Description	2.1
3.0	Slug	g Test I	Response and Analysis	3.1
4.0	Slug	g-Test]	Results	4.1
	4.1	Well	399-2-5 (C5708)	4.2
		4.1.1	Zone 1 (Depth: 10.1 to 10.9 m)	4.2
		4.1.2	Zone 2 (Depth: 21.3 to 22.3 m)	4.5
		4.1.3	Zone 3 (Depth: 37.5 to 38.1 m)	4.8
	4.2	Well	399-3-22 (C5706)	4.9
		4.2.1	Zone 1 (Depth: 16.3 to 16.7 m)	4.10
		4.2.2	Zone 2 (Depth: 23.6 to 24.3 m)	4.12
		4.2.3	Zone 3 (Depth: 39.0 to 41.1 m)	4.13
		4.2.4	Zone 4 (Final Well-Screen Section, Depth: 38.2 to 41.1 m)	4.15
	4.3	Well	399-4-14 (C5707)	4.16
		4.3.1	Zone 1 (Depth: 14.5 to 15.6 m)	4.17
		4.3.2	Zone 2 (Depth: 24.5 to 25.4 m)	4.19
		4.3.3	Zone 3 (Depth: 31.9 to 32.7 m)	4.19
		4.3.4	Zone 4 (Depth: 36.6 to 39.1 m)	4.20
5.0	Hyd	raulic	Conductivity Depth Profile	5.1
6.0	Con	clusior	18	6.1
7.0	Refe	erences	·	7.1
App	endix	A - S	lug Test Field Notes for Wells 399-2-5, 399-3-22, and 399-4-14	A.1
Арр	endix	к B – В	orehole Logs for Wells 399-2-5, 399-3-22, and 399-4-14	B .1

Figures

1.1	Map Showing Locations of 300 Area VOC Well Sites	1.2
2.1	General Slug-Test Configuration within the Temporary Test-Casing String during the Drilling and Testing of Wells 399-2-5, 399-3-22, and 399-4-14	2.3
2.2	General Slug-Test Configuration within the Final Well-Screen Section	2.4
3.1	Diagnostic Slug Test Response	3.2
3.2	Over-Damped Slug-Test Response as a Function of Test-Interval Hydraulic Conductivity	3.3
4.1	Example of Slug-Withdrawal Test Pressure Response for Zone 1, Well 399-2-5	4.4
4.2	Over-Damped Slug-Test Type-Curve Plots for Various High K Values With Zone 1 Test Parameters, Well 399-2-5	4.5
4.3	Selected Diagnostic Plot for Zone 2, Well 399-2-5	4.6
4.4	Selected Type-Curve Analysis Plots, Zone 2, Well 399-2-5, for the Artificially Created, High-Permeability Inner-Zone and the Lower Permeability Outer-Zone Formation	4.7
4.5	Slug Test Pressure Response and Time History Match for Zone 3, Well 399-2-5	4.9
4.6	Slug Test Pressure Response for Zone 3, Well 399-2-5 and Time History Sensitivity Analysis Match for Varying K Values	4.10
4.7	Example of Test-Configuration Induced Oscillatory Response for Zone 1, Well 399-3-22	4.11
4.8	Over-Damped Slug-Test Type-Curve Plots for Various High K Values With Zone 1 Test Parameters, Well 399-3-22	4.12
4.9	Selected Diagnostic Plot for Zone 2, Well 399-3-22	4.14
4.10	Selected Type-Curve Analysis Plots, Zone 2, Well 399-3-22, for the Artificially Created, High-Permeability Inner-Zone and the Lower Permeability Outer-Zone Formation	4.14
4.11	Selected Type-Curve Analysis Plot for Zone 4, Well 399-3-22	4.16
4.12	Example of Test-Configuration Induced Oscillatory Response for Zone 1, Well 399-4-14	4.18
4.13	Over-Damped Slug-Test Type-Curve Plots for Various High K Values With Zone 1 Test Parameters, Well 399-4-14	4.18
4.14	Selected Type-Curve Analysis Plot for Zone 3, Well 399-4-14	4.20
4.15	Selected Type-Curve Analysis Plot for Zone 4, Well 399-4-14	4.21
5.1	Hydraulic Conductivity Depth Profile for Well 399-2-5	5.1
5.2	Hydraulic Conductivity Depth Profile for Well 399-3-22	5.2
5.3	Hydraulic Conductivity Depth Profile for Well 399-4-14	5.2

Tables

4.1	Slug-Test Characteristics for Selected Test/Depth Intervals at 300 Area VOC Test	
	Wells 399-2-5, 399-3-22, and 399-4-14	4.1
4.2	Slug Test Analysis Results for Wells 399-2-5, 399-3-22, and 399-4-14	4.3

1.0 Introduction

Pacific Northwest National Laboratory conducted multiple, stress-level slug tests at selected test/ depth intervals within wells 399-2-5, 399-3-22, and 399-4-14 as part of the 300 Area volatile organic compound (VOC) characterization program at the Hanford Site in Washington State for the U.S. Department of Energy (Figure 1.1). The temporary test screen lengths were characterized as the borehole was advanced to its final drill depth and before its completion as a monitor-well facility. Where possible, the final well-screen sections were characterized following well completion. The primary objective of the slug tests was to provide information pertaining to the vertical distribution of hydraulic conductivity with depth at these locations and to select the final well screen-depth interval for each monitor-well facility. This type of characterization information is important for predicting/simulating contaminant migration (i.e., numerical flow/transport modeling) and designing proper monitor-well strategies within this area.

Section 2 describes the general hydrologic test system employed to perform the series of multiple, stress-level slug tests for each isolated test-interval section. Section 3 discusses slug-test response and analysis methods. Section 4 presents pertinent information describing slug-testing activities and analysis results for the test/depth zones that were hydrologically characterized at the 300 Area VOC wells. Slug-test results are described for each individual test zone within each of the three well locations. Section 5 presents the hydraulic conductivity depth profiles obtained at each of the three well sites. Conclusions and references are provided in Sections 6 and 7, respectively. Slug-test field notes are provided in Appendix A, and borehole lithologic logs are presented in Appendix B.



Figure 1.1. Map Showing Locations of 300 Area VOC Well Sites

2.0 Hydrologic Test System Description

The following discussion of the general hydrologic test plan is taken primarily from similar slug-test characterization-program descriptions presented previously by Spane.^(a) Hydrologic testing was implemented when the approximate targeted depth intervals within the unconfined aquifer were reached during drilling. To prepare the test zone for slug-test characterization, the packer/well-screen test assembly was lowered to the bottom of the borehole, and the drill casing was retracted, exposing an open borehole section of \sim 1 m or less within the Hanford formation and between 1 and 3.5 m within the Ringold Formation. The packer was then inflated to isolate the well-screened/test interval and the testing string from the inside of the drill casing. Following well completion, slug-test characterization was conducted in the final, completed well, well-screen section.

A series of multiple, stress-level slug tests were attempted for each isolated test-interval section. The reason for using a multi-stress-level approach was to determine whether the associated slug-test responses exhibited either a variable or stress-level dependence. As noted in Butler (1998) and Spane et al. (2003b), tests exhibiting either variable or stress-level dependence can provide valuable information pertaining to the presence of dynamic well skin or non-linear (i.e., turbulence) test-response conditions occurring within the test section. General slug-test stress levels applied during testing were designed to be within the range of ~0.3 to 0.7 m for lower stress tests and ~0.7 to 1.4 m for higher stress tests. The slug tests were initiated with two slugging rods of different, known displacement volumes. For most test zones, three or more multi-stress slug tests were conducted. Efforts were made to allow individual slug tests to approach full recovery before starting the next slug test within the characterization sequence. A wide range in recovery times was expected based on the anticipated range in permeability conditions. For example, Spane et al. (2001a, 2001b, 2002, 2003a) and Spane and Newcomer (2004) report recovery times as rapid as <15 sec for high-permeability test intervals (e.g., Hanford formation) to >10 min for lower permeability Ringold Formation test zones. A description of the hydrologic test system used during slug test characterization is provided in the following report section.

Figure 2.1 shows the general test-system configuration used for the slug tests conducted during the drilling and testing of 300 Area wells 399-2-5, 399-3-22, and 399-4-14. Slug tests were conducted with only slugging rods for all test zones (i.e., no pneumatic slug tests were performed). The test-system configuration includes a downhole inflatable packer/well-screen test assembly, a downhole pressure

Spane FA, Jr. 2005b. *Slug Test Characterization Results for Multi-Test/Depth Intervals Conducted During the Drilling of CERCLA Operable Unit UP-1 Wells 299-W19-48, 699-30-66, and 699-36-70B*. Letter report to Mark Byrnes (Fluor-Hanford, ORP), September 13, 2005.

Spane FA, Jr. 2005c. *Slug Test Characterization Results for Multi-Test/Depth Intervals Conducted During the Drilling of CERCLA Operable Unit ZP-1 Wells 299-W11-43, 299-W15-50, and 299-W18-16*. Letter report to Mark Byrnes (Fluor-Hanford, ORP), September 13, 2005.

⁽a) FA Spane, Jr. 2003. Slug Test Characterization Results for Multi- Test/Depth Intervals Conducted During the Drilling of WMA-C Well 299-E27-22 (C4124). Letter report to Jane Borghese (Fluor Hanford, Inc.), October 8, 2003.

Spane FA, Jr. 2005a. *Slug Test Characterization Results for Multi-Test/Depth Intervals Conducted During the Drilling of WMA-BX-BY Well 299-E33-49*. Letter report to Jane Borghese (Fluor-Hanford, ORP), January 10, 2005.

transducer, a slugging rod lowered by a drill rig, and a surface data-logger system. The drill-casing string used for borehole advancement during the drilling of each well had inside diameter (I.D.) and outside diameter (O.D.) dimensions of 0.273 m (10 ³/₄ in.) and 0.298 m (11 ³/₄ in.), respectively. Except for slug tests conducted in Zone 1 near the water table, an inflatable packer was used to seal and isolate the temporary test screen length and test-casing string from the encompassing drill-casing area. The packer was not inflated during testing near the water table. While the packer was inflated, test-interval isolation was verified by adding ~20 L of water above the packer (i.e., in the annular area between the testing string and drill casing), both at the beginning and end of the testing sequence. A 3.05-m length of 0.152-m I.D., 30-slot (Schedule 20), well-screen section attached below the packer was used to maintain an open section for testing the formation after retracting the drill casing. A 0.1-m-long cap was attached to the bottom of the well-screen section. The inside diameter of the attached test-casing string above the well-screen section was 0.102 m. A Druck, Inc. pressure transducer strain-gauge, 0- to 69-kPa (0- to 10-psig) pressure transducer was installed below the fluid-column surface within the temporary test-casing string to monitor downhole test-interval response before and during slug testing. Pressure-transducer measurements were recorded with a Campbell Scientific, Inc. model CR-10X[™] data logger.

Figure 2.2 shows the general slug-test configuration following well completion of each of the three wells. Slug tests were conducted in the final well-screen sections with the same slug rods as those used during testing within the temporary test-casing string. The test-system configuration within the final well-screen section includes a downhole pressure transducer, a slugging rod lowered by a drill rig, and a surface data-logger system. The final 10-slot (Schedule 10) well-screen section had a length of 3.0 m (10 ft) or 6.1 m (20 ft) and an I.D. dimension of 0.152 m (6 in.). A Druck, Inc. pressure transducer strain-gauge, 0- to 69-kPa (0- to 10-psig) pressure transducer was installed below the fluid-column surface within the well casing or well-screen section to monitor downhole test-interval response before and during slug testing. Pressure-transducer measurements were recorded with a Campbell Scientific, Inc. model CR-10X[™] data logger.



Figure 2.1. General Slug-Test Configuration within the Temporary Test-Casing String during the Drilling and Testing of Wells 399-2-5, 399-3-22, and 399-4-14



Figure 2.2. General Slug-Test Configuration within the Final Well-Screen Section

3.0 Slug Test Response and Analysis

The following discussion pertaining to slug-test response and analysis is taken primarily from Spane [see footnote 2.0 (a)]. As shown in Figure 3.1 and discussed in Butler (1998) and Spane et al. (2003b), water levels within a test well can respond in one of three ways to the instantaneously applied stress of a slug test. These response model patterns are 1) an over-damped response, where the water levels recover in an exponentially decreasing recovery pattern, 2) an under-damped response, where the slug-test response oscillates above and below the initial static, with decreasing peak amplitudes with time, and 3) a critically damped response, where the slug test behavior exhibits characteristics that are transitional to the over- and under-damped response patterns. Factors that control the type of slug-test response model that is exhibited within a well include a number of aquifer properties (hydraulic conductivity) and well-dimension characteristics (well-screen length, well-casing radius, well-radius, aquifer thickness, fluid-column length) and can be expressed by the response-damping parameter, C_D, which Butler (1998) reports for unconfined aquifer tests as:

$$C_{D} = \sqrt{\frac{g}{L_{e}}} * \frac{r_{c}^{2} \ln[R_{e}/r_{w}]}{2KL}$$
(3.1)

where g = acceleration due to gravity

- $L_e = effective well water-column length$
- r_c = well casing radius; i.e., radius of well water-column that is active during testing
- R_e = effective test radius parameter; as defined by Bouwer and Rice (1976)
- $r_w =$ well radius
- K = hydraulic conductivity of test interval
- L = well-screen length.



Figure 3.1. Diagnostic Slug Test Response (taken from Spane and Newcomer 2008)

Given the multitude of possible combinations of aquifer properties, well-casing dimensions, and testinterval lengths, no universal C_D value ranges can be provided that describe slug-test response conditions. However, for various combinations anticipated for testing at 300 Area VOC well sites during drilling, the following general guidelines on predicting slug-test responses are provided:

- $C_D > 3 = \text{over-damped response}$
- $C_D 1 3 =$ critically damped response
- C_D <1 = under-damped response.

An over-damped test response generally occurs within stress wells monitoring test formations of low to moderately high hydraulic conductivity (e.g., Ringold Formation) and are indicative of test conditions where frictional forces (i.e., resistance of groundwater flow from the test interval to the well) are predominant over test-system inertial forces. Figure 3.2 shows predicted slug-test recovery as a function of hydraulic conductivity (K range: 1.0 to 40 m/day; 1.0-m test interval) for test intervals exhibiting over-damped response characteristics and for general 300 Area VOC test well/interval conditions. The test predictions shown in the figure are based on responses occurring within a test system casing I.D. = 0.1016 m. As indicated in the figure, test intervals having hydraulic conductivity values of approximately 40 m/day or less should be readily resolved for tests exhibiting over-damped slug-test behavior. For over-damped slug tests, two different methods can be used for the slug-test analysis: the semi-empirical, straight-line analysis method described in Bouwer and Rice (1976) and Bouwer (1989) and the type-curve-matching method for unconfined aquifers presented in Butler (1998). For

over-damped slug tests, hydraulic-conductivity estimates obtained with the Bouwer and Rice analytical method are generally less reliable than corresponding estimates obtained with the type-curve-matching method (Hyder and Butler 1995; Butler 1998). For this reason, only the type-curve-matching analytical method was used for estimating hydraulic conductivity for zones tested at the 300 Area VOC wells. A detailed description of over-damped, slug-test-analysis methods is presented in Spane and Newcomer (2004).



Figure 3.2. Over-Damped Slug-Test Response as a Function of Test-Interval Hydraulic Conductivity

The time-history matching method is used for lower permeability test intervals, where individual over-damped slug tests do not fully recover to pre-test conditions. This analysis method is based on the superposition principle that relies on super-imposing the predicted slug-test responses of subsequent tests that are conducted in a series. The predicted slug-test responses are calculated with the type-curve matching method described in Butler (1998), which are combined by super-imposing their individual responses for the respective times of test initiation to yield a predicted composite test response (i.e., time-history response). The analysis method is greatly facilitated by maintaining uniform slug-test recovery time periods and using equal slug-test volume displacements (e.g., alternating slug withdrawal and injection tests with a slugging rod, each phase for 30-minute periods). Strictly speaking, the super-position principle (and time-history matching) is only appropriate for linear-response aquifer systems (e.g., confined aquifers). However, Reilly et al. (1987) state that it is also appropriate for unconfined aquifer's thickness. This was the case for the time-history analysis of the testing sequence conducted for well 399-2-5, Zone 3.

Under-damped test-response patterns are exhibited within stress wells where inertial forces are predominant over formational frictional forces. This commonly occurs in wells with extremely long fluid columns (i.e., large water mass within the well column) and/or that penetrate highly permeable aquifers (e.g., Hanford formation). Tests exhibiting under-damped behavior should be conducted with very small stress-level applications. No 300 Area VOC well test intervals displayed formational test-response characteristics that were under damped.

As mentioned previously, critically damped test responses are indicated by stress well water-level responses that are transitional to the over- and under-damped test conditions, as shown in Figure 3.1. They typically occur in wells that monitor test formations exhibiting intermediate to high hydraulic conductivity. As noted in Butler (1998), distinguishing between slug-test responses that are over damped and critically damped may be difficult in some cases (i.e., due to test signal noise) when examined on arithmetic plots. Proper model identification may be enhanced when semi-log plots are used, i.e., log head versus time (e.g., Bouwer and Rice plot). Critically damped slug tests exhibit a diagnostic concavedownward pattern when plotted in this semi-log plot format. This is in contrast to over-damped response behavior, which displays either a linear or concave upward (elastic) pattern. Critically damped slug-test responses are influenced by processes (e.g., inertial) that are not accounted for in the previously discussed slug-test analytical methods (i.e., for over-damped tests). Because of this, slug tests exhibiting these response characteristics cannot be analyzed quantitatively with the Bouwer and Rice or standard typecurve methods. High-K analysis methods that can be employed for analyzing unconfined aquifer tests exhibiting response behavior that is either critically damped or under damped include those described in Springer and Gelhar (1991), Butler (1998), McElwee and Zenner (1998), McElwee (2001), Butler and Garnett (2000), and Zurbuchen et al. (2002). Because of the ease provided by a spreadsheet-based approach, the test-analysis method presented in Butler and Garnett (2000) is preferred for analyzing tests exhibiting critically damped behavior. A detailed discussion of this analytical procedure and method is presented in Spane and Newcomer (2004). No 300 Area VOC well test intervals displayed formational test-response characteristics that were critically damped.

4.0 Slug-Test Results

The following discussion presents pertinent information describing slug-testing activities and analysis results for the test/depth zones that were hydrologically characterized at the 300 Area VOC boreholes as they were advanced to their final drilling depths. Table 4.1 presents slug-test information for the respective test/depth intervals, while Table 4.2 summarizes the slug-test-analysis results. Selected borehole logs are presented in Appendix B, which can be referred to for a geologic description of the respective well test zone/depth intervals.

			Test P	arameters			
Test Well Number (Borehole ID)	Test Zone	Test Date	Number of Slug Tests	Depth to Water (m bgs)	Depth/Test Interval (m bgs)	Diagnostic Slug- Test Response Model	Hydrogeologic Unit Tested ^(a)
	Zone 1	9/7/07	4	9.72	10.1–10.9 (0.8)	Homogeneous Formation/ Exponential-Decay (over-damped)	Hanford formation (Unit 1)
399-2-5 (C5708)	Zone 2	9/19/07	8	10.09	21.3–22.3 (1.0)	Heterogeneous Formation/ Exponential-Decay (over-damped)	Ringold Formation (Unit 5)
	Zone 3	9/27/07	1	9.69	37.5–38.1 (0.6)	Homogeneous Formation/ Exponential-Decay (over-damped)	Ringold Formation (Unit 5)
	Zone 1	11/6/07	9	13.50	16.3–16.7 (0.4)	Homogeneous Formation/ Exponential-Decay (over-damped)	Hanford formation (Unit 1)
399-3-22	Zone 2	11/8/07	3	13.55	23.6–24.3 (0.7)	Heterogeneous Formation/ Exponential-Decay (over-damped)	Ringold Formation (Unit 5)
(C5706)	Zone 3	11/15/07	0	13.93	39.0–41.0 (2.0)	NA	Ringold Formation (Unit 5)
	Zone 4 ^(b)	12/3/07	6	13.29	38.2–41.1 (2.9)	Homogeneous Formation/ Exponential-Decay (over-damped)	Ringold Formation (Unit 5)

Table 4.1.Slug-Test Characteristics for Selected Test/Depth Intervals at 300 Area VOC Test Wells399-2-5, 399-3-22, and 399-4-14

Table 4.1. (Cont'd.)

			Test P	arameters			
Test Well Number (Borehole ID)	Test Zone	Test Date	Number of Slug Tests	Depth to Water (m bgs)	Depth/Test Interval (m bgs)	Diagnostic Slug- Test Response Model	Hydrogeologic Unit Tested ^(a)
	Zone 1	10/10/07	8	13.05	14.5–15.6 (1.1)	Homogeneous Formation/ Exponential-Decay (over-damped)	Hanford formation (Unit 1)
399-4-14	Zone 2	10/15/07	0	13.03	24.5–25.4 (0.9)	NA	Ringold Formation (Unit 5)
(C5707)	Zone 3	10/19/07	6	13.14	31.9–32.7 (0.8)	Homogeneous Formation/ Exponential-Decay (over-damped)	Ringold Formation (Unit 5)
	Zone 4	10/23/07	6 ^(c)	13.01	36.6–39.1 (2.5)	Homogeneous Formation/ Exponential-Decay (over-damped)	Ringold Formation (Unit 5)

NA = Not applicable.

Note: For all test wells, r_c ranged between 0.0508 and 0.0762 meter; $r_w = 0.1492$ meters. Hydrogeologic unit number in parentheses indicates the relevant groundwater-flow model layer, as described in Thorne et al. (1993).

(a) Assumed to be uniform within the well-screen test section.

(b) Final, completed well-screen section.

(c) No quantitative analysis was possible for one of the slug tests because the response of the exponential decay test was disturbed.

4.1 Well 399-2-5 (C5708)

The drilling of 300 Area VOC well 399-2-5 was initiated on September 4, 2007, and continued until reaching a final depth of 39.8 m bgs on September 28, 2007. The Lower Mud unit of the Ringold Formation was encountered at a depth of 38.1 m bgs, which represents the bottom boundary of the unconfined aquifer at this location. Three test-depth intervals were tested at the borehole location: Zone 1 = 10.1 to 10.9 m bgs; Zone 2 = 21.3 to 22.3 m bgs; and Zone 3 = 37.5 to 38.1 m bgs. Slug tests conducted within the final, completed well-screen section, with a test/depth interval of 9.9 to 12.5 m bgs, yielded test results similar to the Zone 1 test results (i.e., full recovery within ~3 seconds after test initiation). Because of these similar test results, the data analysis for tests conducted within the completed well-screen section was not included in this report. The slug-test field notes for this test/depth interval, however, are provided in Appendix A.

4.1.1 Zone 1 (Depth: 10.1 to 10.9 m)

After reaching a drill depth of 11.3 m bgs, the bottom 0.3 m of the borehole filled in with sediment slough. The well-screen assembly was lowered to the bottom of the borehole at a depth of 11.0 m bgs, and the 0.2985-m O.D. (11 ³/₄-in. O.D.) drill casing was retracted 0.9 m (i.e., from 11.0 to 10.1 m bgs),

producing a test/depth interval for Zone 1 of 10.1 to 10.9 m bgs (bottom end-cap at 10.9 to 11.0 m bgs). The borehole geology log (Appendix B; Figure B.1) indicates that the test-interval section generally consists of a sandy gravel and gravel unit, composed of \sim 80% gravel, \sim 15% sand, and \sim 5% silt. At the time of testing, the well-screen test interval was located ~0.4 m below the unconfined aquifer water-table surface, and test results reflect sediments of the Hanford formation (Unit 1).

		Type-Curve An	alysis Method
Test Well Number (Borehole ID)	Test Zone	Hydraulic Conductivity, K _h , ^(a) (m/day)	Specific Storage, $S_s(m^{-1})$
	Zone 1	≥300	1.0E-5
399-2-5	Zone 2 (outer zone formation)	1.17–1.73 (1.42)	1.0E-5
(C5708)	Zone 2 (artificially- created inner zone)	6.05–9.50 (7.72)	1.0E-5
	Zone 3	≤0.01	1.0E-5
	Zone 1	≥400	1.0E-5
	Zone 2 (outer zone formation)	0.32–0.61 (0.44)	1.0E-5
399-3-22 (C5706)	Zone 2 (artificially- created inner zone)	1.56	1.0E-5
	Zone 3	NA	NA
	Zone 4 ^(b)	1.04–1.51 (1.34)	1.0E-5
	Zone 1	≥300	1.0E-5
	Zone 2	NA	NA
399-4-14 (C5707)	Zone 3	2.20–2.85 (2.48)	7.0E-4–1.6E-3
	Zone 4	0.93–1.12 (1.04)	1.0E-4-2.0E-4
NA = Not applicable. Note: Number in parenth (a) Assumed to be unifor (b) Final, completed wel	rm within the well-screen		

Table 4.2. Slug Test Analysis Results for Wells 399-2-5, 399-3-22, and 399-4-14

A series of four slug withdrawal tests (one low-stress and three high-stress tests) were conducted between 0742 hours and 0810 hours (Pacific Standard Time [PST]), September 7, 2007. The slug tests were conducted with two different sized slugging rods that were partially submerged in the water column, one with a partially submerged volume of 0.0027 m³ and a larger one with a partially submerged volume of 0.0059 m³. These partially submerged slug-rod volumes imparted a theoretical applied stress level of 0.15 m for the low-stress test and 0.32 m for the high-stress tests within the 0.1524-m (6-in.) I.D. temporary screen. Downhole test-interval response pressures during testing were monitored with a 0- to 69-kPa (0- to 10-psig) pressure transducer set at a depth of ~11-m bgs. The static depth-to-water for the test interval measured before testing was 9.72 m bgs.

All slug tests for Zone 1 exhibited full recovery within ~3 seconds after test initiation. A selected example of the pressure-test response for one of the slug-withdrawal tests is shown in Figure 4.1. Test responses of the formation dissipated within the initial seconds of the test because of very high-permeability test conditions within the Hanford formation. This rapid test response (i.e., 90% recovery within ~3 seconds) will be used for analyzing a minimum value for K.



Figure 4.1. Example of Slug-Withdrawal Test Pressure Response for Zone 1, Well 399-2-5

To provide a bounding, greater-than K estimate for Zone 1 tests, a series of over-damped slug-test type curves were generated for various high K values with Zone 1 test parameters. Normalized plots of the type curves for K values ranging between 100 and 1000 m/day are shown in Figure 4.2. The plots in Figure 4.2 indicate a range of $K \ge 300$ m/day (and assuming $S_s = 1.0E-05$ m⁻¹) that correspond to rapid, formational test response recovery of 90% within ~3 seconds. This lower bounding limit value for K is the best estimate available for the Zone 1 test/depth interval.



Figure 4.2. Over-Damped Slug-Test Type-Curve Plots for Various High K Values With Zone 1 Test Parameters, Well 399-2-5

4.1.2 Zone 2 (Depth: 21.3 to 22.3 m)

After reaching a drill depth of 22.9 m bgs, the bottom 0.5 m of the borehole filled in with sediment slough. The packer/well-screen assembly was lowered to the bottom of the borehole at a depth of 22.4 m bgs, and the 0.2985-m O.D. (11 ³/₄-in. O.D.) drill casing was retracted 1.1 m (i.e., from 22.4 to 21.3 m bgs), producing a test/depth interval for Zone 2 of 21.3 to 22.3 m bgs (bottom end-cap at 22.3 to 22.4 m bgs). The borehole geology log (Appendix B; Figure B.1) indicates that the test-interval section generally consists of a sandy gravel unit, composed of ~60% gravel and ~40% sand. At the time of testing, the well-screen test interval was located ~11.2 m below the unconfined aquifer water-table surface, and test results reflect sediments of the Ringold Formation (Unit 5).

A series of four slug-injection tests and four slug-withdrawal tests (four low-stress and four highstress tests) were conducted between 0931 hours and 1252 hours (PST), September 19, 2007. The slug tests were conducted with two different-sized slugging rods, one with a volume of 0.0055 m³ and a larger one with a volume of 0.011 m³. These slug-rod volumes imparted a theoretical applied stress level of 0.68 m for the low-stress tests and 1.36 m for the high-stress tests within the 0.1016-m (4-in.) I.D. temporary casing string. Downhole test-interval response pressures during testing were monitored with a 0- to 69-kPa (0- to 10-psig) pressure transducer set at a depth of ~13-m bgs. The static depth-to-water for the test interval measured before testing was 10.09 m bgs. A diagnostic analysis of slug tests conducted for this test/depth interval indicates a heterogeneous formation response condition. This test pattern exhibits a high-permeability, inner-zone response during the initial fast-recovery portion of the test that slowly transitions to a lower permeability response for the surrounding outer-zone formation. The presence of an elastic, high-permeability inner-zone reflects an artificially induced condition that was likely attributed to collapse of unconsolidated formation sediments around the temporary well screen as the drill casing was retracted. An examination of the drilling log geologic description indicates sand heaving within this test interval during drilling.

As discussed in Spane (1993), slug tests exhibiting linear response characteristics for heterogeneous formation tests can be analyzed using the homogeneous formation analysis approaches described in Section 3.0. A comparison of the normalized, higher and lower stress, slug-test responses indicated identical behavior. For the homogeneous-formation analysis, the type-curve method estimates for K ranged between 1.17 and 1.73 m/day (average of 1.42 m/day) for the outer-zone formation and ranged between 6.05 and 9.50 m/day (average of 7.72 m/day) for the artificially created, higher permeability, inner-zone. Selected examples of the diagnostic and test-analysis plots for this test/depth interval are shown in Figure 4.3 and Figure 4.4(a, b), respectively.



Figure 4.3. Selected Diagnostic Plot for Zone 2, Well 399-2-5



Figure 4.4. Selected Type-Curve Analysis Plots, Zone 2, Well 399-2-5, for (a) the Artificially Created, High-Permeability Inner-Zone and (b) the Lower Permeability Outer-Zone Formation

4.1.3 Zone 3 (Depth: 37.5 to 38.1 m)

After driving the 0.2985-m O.D. (11 ³/₄-in. O.D.) drill casing to a depth of 37.5 m bgs and then drilling the open hole to a depth of 38.7 m bgs, the packer/well-screen assembly was lowered to the bottom of the borehole, exposing the temporary screen to the formation at a depth interval of 37.5 to 38.6 m bgs (bottom end-cap at 38.6 to 38.7 m bgs). The depth/test interval for Zone 3 was drilled ahead of the drill casing because, unlike the test/depth intervals for Zones 1 and 2 for this borehole, the formation sediments were consolidated, and the borehole remained open during drilling. While pumping during groundwater sampling before conducting the slug tests, the bottom 0.5 m of the well-screen section filled in with sediment slough (i.e., fine-grained sediments), effectively reducing the test/depth interval for Zone 3 to 37.5 to 38.1 m bgs. The borehole geology log (Appendix B; Figure B.1) indicates that the test-interval section generally consists of a silty sandy gravel unit, composed of 50 to 75% gravel, 25 to 50% sand, and <10% silt. At the time of testing, the well-screen test interval was located ~27.8 m below the unconfined aquifer water-table surface, and test results reflect sediments of the Ringold Formation (Unit 5) just above the Ringold Lower Mud unit.

A series of three slug withdrawal and injection tests were conducted between 0843 hours and 0943 hours (PST), September 28, 2007. The tests were initiated by rapidly withdrawing a 0.0055-m³ volume slugging rod (slug withdrawal test) from the fluid column within the 0.102-m (4-in.) I.D. testingstring casing used to set the packer/well screen assembly. After 30 minutes of recovery, the slugging rod was rapidly immersed into the fluid column initiating a slug-injection test. After another 30 minutes of recovery (from the slug-injection test), a third slug test (slug withdrawal test) was initiated by rapidly withdrawing the slugging rod from the fluid column. Downhole test-interval response pressures during testing were monitored with a 0- to 69-kPa (0- to 10-psig) pressure transducer set at a depth of ~12 m bgs. The depth-to-water for the test interval measured before testing was 9.69 m bgs. This depth-to-water level is not representative of "static" conditions since a declining water-level trend of -0.0008104 m/min was observed for an extended period before and during slug testing.

The three slug tests indicated a recovery response behavior that was extremely slow and overdamped. Since the recovery times for the slug tests were slow, with <10% recovery of the applied stress, a test history match approach was used for analyzing the slug withdrawal and injection phases of the tests. Since a declining water-level trend (i.e., -0.0008104 m/min) was observed during the test phases, the predicted test responses were super-imposed on this trend to match the observed test responses.

Figure 4.5 shows the observed responses for each of the three slug-test phases and the predicted time history match for the testing sequence. As noted previously, a declining water-level trend of -0.0008104 m/min was observed over the test period. As indicated in Figure 4.5, a hydraulic conductivity K estimate of 0.01 m/day provides a good match to the observed test-response sequence.

To demonstrate the sensitivity of the analytical solution, Figure 4.6 shows the predicted test history match with K values of 0.01, 0.02, 0.05, and 0.1 m/day. As indicated, test-history matches depart significantly from the observed test responses with K values higher than 0.01 m/day. This suggests a K value of ≤ 0.01 m/day for the depth interval tested. Since slough filled in the lower 0.5 m of the well-screen section before slug testing, there is some uncertainty whether the relatively low hydraulic conductivity indicated for Zone 3 is representative of *in situ* formation conditions or is an artifact of borehole instability around the well screen. Pumping for groundwater samples from the test interval before slug testing may have contributed to this instability. However, the low-permeability condition is corroborated by the

proximity of the test interval to the underlying Lower Mud unit of the Ringold Formation and by the observed slow static water-level recovery trend before and during the slug tests.

4.2 Well 399-3-22 (C5706)

The drilling of 300 Area VOC well 399-3-22 was initiated on October 31, 2007, and continued until reaching a final depth of 42.8 m bgs on November 13, 2007. The Lower Mud unit of the Ringold Formation was encountered at a depth of 41.1 m bgs, which represents the bottom boundary of the unconfined aquifer at this location. Three test-depth intervals were successfully tested at the borehole location: Zone 1 = 16.3 to 16.7 m bgs; Zone 2 = 23.6 to 24.3 m bgs; and Zone 4 = 38.2 to 41.1 m bgs. Zone 4 represents the final well-screen completion. One projected test-depth interval during drill advancement, Zone 3 = 39.0 to 41.0 m bgs, was not tested.



Figure 4.5. Slug Test Pressure Response and Time History Match for Zone 3, Well 399-2-5



Figure 4.6. Slug Test Pressure Response for Zone 3, Well 399-2-5 and Time History Sensitivity Analysis Match for Varying K Values

4.2.1 Zone 1 (Depth: 16.3 to 16.7 m)

After reaching a drill depth of 16.8 m bgs, the well-screen assembly was lowered to the bottom of the borehole, and the 0.2985-m O.D. (11 $\frac{3}{4}$ -in. O.D.) drill casing was retracted 0.5 m, producing a test/depth interval for Zone 1 of 16.3 to 16.7 m bgs (bottom end-cap at 16.7 to 16.8 m bgs). The borehole geology log (Appendix B; Figure B.2) indicates that the test-interval section generally consists of a sandy gravel unit, composed of >60% gravel, <40% sand, and <1% silt. At the time of testing, the well-screen test interval was located ~2.8 m below the unconfined aquifer water-table surface, and test results reflect sediments of the Hanford formation (Unit 1).

A series of five slug-injection tests and four slug-withdrawal tests (five low-stress and four highstress tests) were conducted between 0632 hours and 0743 hours (PST), November 6, 2007. The slug tests were conducted with two different-sized slugging rods, one with a volume of 0.0055 m³ and a larger one with a volume of 0.011 m³. These slug-rod volumes imparted a theoretical applied stress level of 0.30 m for the low-stress tests and 0.61 m for the high-stress tests within the 0.1524-m (6-in.) I.D. temporary well-screen. For the two high-stress slug-withdrawal tests, the pressure transducer cable was attached to the slugging rod to prevent the rod and transducer probe from becoming lodged inside the 0.1016-m (4-in.) I.D. casing above the screen during slug withdrawal. Downhole test-interval response pressures during testing were monitored with a 0- to 69-kPa (0- to 10-psig) pressure transducer set at a depth of ~16-m bgs for the first test and ~14-m bgs for the remaining slug tests. The static depth-to-water for the test interval measured before testing was 13.50 m bgs.

All slug tests for Zone 1 exhibited oscillations attributed to a test-configuration-induced condition. The oscillatory test pattern is believed to be primarily due to pressure imbalances between the water column inside the temporary well-screen and the water column in the annular space between the well-screen and drill casing. A selected example of these test-induced oscillations is shown in Figure 4.7. The oscillations in Figure 4.7 indicate a pressure change, ~0.04 m, immediately after test initiation that is significantly less than the theoretical H_0 value of 0.30 m for this low-stress test (than would occur only within the well screen). This is an indication that the oscillatory test response. The actual applied stress, H_0 , to the formation is uncertain due to the time it takes (i.e., 1 to 2 seconds) to completely remove the slug rod and recover the associated rapid test response. Due to very high-permeability test conditions within the Hanford formation, formational test responses dissipated within the initial seconds (i.e., 90% recovery within ~3 seconds) of the test and are not discernable in the test-configuration-induced oscillatory pattern.



Figure 4.7. Example of Test-Configuration Induced Oscillatory Response for Zone 1, Well 399-3-22

To provide a bounding, greater-than K estimate for the Zone 1 tests, a series of over-damped slug-test type curves were generated for various high K values, using Zone 1 test parameters. Normalized plots of the type curves for K values ranging between 100 and 1000 m/day are shown in Figure 4.8. The plots in Figure 4.8 indicate a range of $K \ge 400$ m/day (and assuming $S_s = 1.0E-05$ m⁻¹) that correspond to rapid, formational test-response recovery of 90% within ~3 seconds. This lower bounding limit value range for K is the best available estimate for the Zone 1 test/depth interval.

4.2.2 Zone 2 (Depth: 23.6 to 24.3 m)

After reaching a drill depth of 26.2 m bgs, the packer/well-screen assembly was lowered to the bottom of the borehole, and the 0.2985-m O.D. (11 ³/₄-in. O.D.) drill casing was retracted 2.6 m, exposing the temporary screen to the formation at a depth interval of 23.6 to 26.1 m bgs (bottom end-cap at 26.1 to 26.2 m bgs). While pumping during groundwater sampling before conducting the slug tests, the bottom 1.8 m of the well-screen section filled in with sediment slough (i.e., fine-grained sediments), effectively reducing the test/depth interval for Zone 2 to 23.6 to 24.3 m bgs. A depth-to-bottom measurement after the second slug test indicated that this test/depth interval remained open during three of the four slug tests performed at this zone. The borehole geology log (Appendix B; Figure B.2) indicates that the test-interval section generally consists of a silty sand unit, composed of ~90% sand and ~10% silt. At the time of testing, the well-screen test interval was located ~10.1 m below the unconfined aquifer water-table surface, and test results reflect sediments of the Ringold Formation (Unit 5).



Figure 4.8. Over-Damped Slug-Test Type-Curve Plots for Various High K Values With Zone 1 Test Parameters, Well 399-3-22

A series of three slug tests (two low-stress and one high-stress test) were conducted between 1259 hours and 1439 hours (PST), November 8, 2007. A fourth slug test was conducted at 0637 hours, November 9, 2007, but was abandoned early in the test because of additional slough filling in the temporary test/screen interval. It is likely that deflating and re-inflating the packer contributed to the additional slough. The slug tests were conducted with two different sized slugging rods, one with a volume of 0.0055 m³ and a larger one with a volume of 0.011 m³. These slug-rod volumes imparted a

theoretical applied stress level of 0.68 m for the low-stress tests and 1.36 m for the high-stress tests within the 0.1016-m (4-in.) I.D. temporary casing string. Downhole test-interval response pressures during testing were monitored with a 0- to 69-kPa (0- to 10-psig) pressure transducer set at a depth of \sim 16.6 m bgs for the first three tests. The static depth-to-water for the test interval measured before testing was 13.55 m bgs.

A diagnostic analysis of slug tests conducted for this test/depth interval indicates a heterogeneous formation response condition. This test pattern exhibits a high-permeability, inner-zone response during the initial fast-recovery portion of the test that slowly transitions to a lower permeability response for the surrounding outer-zone formation. The presence of an elastic, high-permeability inner-zone reflects an artificially induced condition that was likely attributed to the collapse of unconsolidated formation sediments around the temporary well screen as the drill casing was retracted. An examination of the drilling log geologic description indicates sand heaving within this test interval during drilling.

As discussed in Spane (1993), slug tests exhibiting linear response characteristics for heterogeneous formation tests can be analyzed with the homogeneous formation analysis approaches described in Section 3.0. A comparison of the normalized, higher and lower stress, slug-test responses indicated stress dependence, with higher stress tests exhibiting a delayed test recovery. For the homogeneous-formation analysis, the type-curve method estimates for K ranged between 0.32 and 0.61 m/day (average of 0.44 m/day) for the outer-zone formation, and the estimate was 1.56 m/day for the artificially created, higher permeability, inner zone. Selected examples of the diagnostic and test-analysis plots for this test/depth interval are shown in Figure 4.9 and Figure 4.10(a, b), respectively.

4.2.3 Zone 3 (Depth: 39.0 to 41.1 m)

After reaching a drill depth of 41.1 m bgs and driving the 0.2985-m O.D. (11 ³/₄-in. O.D.) drill casing to a depth of 39.0 m bgs, the packer/well-screen assembly was lowered to the bottom of the borehole, producing a test/depth interval for Zone 3 of 39.0 to 41.0 m bgs (bottom end-cap at 41.0 to 41.1 m bgs). The depth/test interval for Zone 3 was drilled ahead of the drill casing because, unlike the test/depth intervals for Zones 1 and 2 for this borehole, the formation sediments were consolidated, and the borehole remained open during drilling. The borehole geology log (Appendix B; Figure B.1) indicates that the test-interval section generally consists of silty sandy gravel, composed of 60 to 70% gravel, 15 to 30% sand, and 10 to 15% silt. The well-screen test interval was located ~25.1 m below the unconfined aquifer water-table surface, and the test/depth interval reflects sediments of the Ringold Formation (Unit 5) that were resting on the Ringold Lower Mud unit encountered at a depth of 41.1 m bgs.

While pumping during groundwater sampling (before conducting the projected slug tests on November 15, 2007), the entire well-screen test/depth interval filled in with fine-grained sediment. Slug tests were not performed at this test/depth interval for Zone 3 because the final, completed well-screen section was to be constructed over the bottom 2.9 m of the Ringold Formation (Unit 5). However, slugtest results for characterizing hydraulic properties are available for the final, completed well-screen section, which has a 0.9-m longer test/depth interval of 38.2 to 41.1 m bgs (see Section 4.2.4 below for Zone 4).



Figure 4.9. Selected Diagnostic Plot for Zone 2, Well 399-3-22



Figure 4.10. Selected Type-Curve Analysis Plots, Zone 2, Well 399-3-22, for (a) the Artificially Created, High-Permeability Inner-Zone and (b) the Lower Permeability Outer-Zone Formation



Figure 4.10. (contd)

4.2.4 Zone 4 (Final Well-Screen Section, Depth: 38.2 to 41.1 m)

A test/depth interval for Zone 4 of 38.2 to 41.1 m bgs represents the final well-screen section following well completion. The borehole geology log (Appendix B; Figure B.2) indicates that the final well-screen test-interval section generally consists of a silty sandy gravel unit, composed of 60 to 75% gravel, 15 to 30% sand, and 10 to 15% silt. At the time of testing, the well-screen test interval was located ~24.9 m below the unconfined aquifer water-table surface, and test results reflect sediments of the Ringold Formation (Unit 5) just above the Ringold Lower Mud unit.

A series of three slug-injection tests and three slug-withdrawal tests (four low-stress and two highstress tests) were conducted between 1355 hours and 1512 hours (PST), December 3, 2007. The slug tests were conducted with two different-sized slugging rods, one with a volume of 0.0055 m³ and a larger one with a volume of 0.011 m³. These slug-rod volumes imparted a theoretical applied stress level of 0.30 m for the low-stress tests and 0.61 m for the high-stress tests within the 0.1524-m (6-in.) I.D. well casing. Downhole test-interval response pressures during testing were monitored with a 0- to 69-kPa (0- to 10-psig) pressure transducer set at a depth of ~16 m bgs. The static depth-to-water for the test interval measured before testing was 13.29 m bgs.

All slug tests exhibited over-damped (exponential-decay response) homogeneous formation behavior, which is indicative of low to moderate permeability test zone conditions. A comparison of the normalized, higher and lower stress, slug-test responses indicated slight stress dependence, with higher stress tests exhibiting a slightly delayed test recovery. Slug tests exhibiting this type of response behavior can be analyzed quantitatively with homogeneous formation analysis approaches, as described in Butler (1998). For the homogeneous formation analysis, the standard type-curve method provided estimates of K ranging between 1.04 to 1.51 m/day, averaging 1.34 m/day, and S_s of 1.0E-5 m⁻¹. A selected example of the test-analysis plots for this test/depth interval is shown in Figure 4.11.

4.3 Well 399-4-14 (C5707)

The drilling of 300 Area VOC well 399-4-14 was initiated on October 8, 2007, and continued until reaching a final depth of 41.5 m bgs on October 24, 2007. The Lower Mud unit of the Ringold Formation was encountered at a depth of 39.5 m bgs, which represents the bottom boundary of the unconfined aquifer at this location. Three test-depth intervals were tested successfully at the borehole location: Zone 1 = 14.5 to 15.6 m bgs; Zone 3 = 31.9 to 32.7 m bgs; and Zone 4 = 36.6 to 39.1 m bgs. One projected test-depth interval, Zone 2 = 24.5 to 25.4 m bgs, was not tested. Slug tests conducted within the final, completed well-screen section, with a test/depth interval of 13.0 to 17.7 m bgs, yielded test results similar to the Zone 1 test results (i.e., full recovery within ~3 seconds after test initiation). Because of these similar test results, the data analysis for tests conducted within the completed well-screen section was not included in this report. The slug-test field notes for this test/depth interval, however, are provided in Appendix A.



Figure 4.11. Selected Type-Curve Analysis Plot for Zone 4, Well 399-3-22

4.3.1 Zone 1 (Depth: 14.5 to 15.6 m)

After reaching a drill depth of 16.2 m bgs, the bottom 0.4 m of the borehole filled in with sediment slough. The well-screen assembly was lowered to the bottom of the borehole at a depth of 15.8 m bgs, and the 0.2985-m O.D. (11 ³/₄-in. O.D.) drill casing was retracted 1.7 m (i.e., from 16.2 to 14.5 m bgs). Following groundwater sampling from this well screen, the well-screen assembly was inadvertently raised 0.1 m during sample pump removal, producing a test/depth interval for Zone 1 of 14.5 to 15.6 m bgs (bottom end-cap at 15.6 to 15.7 m bgs). The borehole geology log (Appendix B; Figure B.1) indicates no sediment sample recovery from a depth of 13.1 to 17.4 m bgs after cleaning out the borehole. A split-spoon sample collected from a depth (i.e., 13.1 to 14.0 m bgs) above the test-interval section indicates a unit composed of gravels with a sand matrix. At the time of testing, the well-screen test interval was located ~1.5 m below the unconfined aquifer water-table surface, and test results reflect sediments of the Hanford formation (Unit 1).

A series of four slug-injection tests and four slug-withdrawal tests (four low-stress and four highstress tests) were conducted between 0933 hours and 1019 hours (PST), October 10, 2007. The slug tests were conducted with two different sized slugging rods, one with a volume of 0.0055 m³ and a larger one with a volume of 0.011 m³. These slug-rod volumes imparted a theoretical applied stress level of 0.30 m for the low-stress tests and 0.61 m for the high-stress tests within the 0.1524-m (6-in.) I.D. temporary screen. Downhole test-interval response pressures during testing were monitored with a 0- to 69-kPa (0to 10-psig) pressure transducer set at a depth of ~16-m bgs for the first slug test and at a depth of ~14 m bgs for the remaining slug tests. The static depth-to-water for the test interval measured before testing was 13.05 m bgs.

All slug tests for Zone 1 exhibited oscillations attributed to a test-configuration-induced condition. The oscillatory test pattern is believed primarily attributable to pressure imbalances between the water column inside the temporary well-screen and the water column in the annular space between the well-screen and drill casing. A selected example of these test-induced oscillations is shown in Figure 4.12. The oscillations in Figure 4.12 indicate a pressure change, ~0.02 m, immediately after test initiation that is significantly less than the theoretical H_0 value of 0.30 m for this low-stress test (than would occur only within the well screen). This is an indication that the oscillatory test response. The actual applied stress, H_0 , to the formation is uncertain due to the time required (i.e., 1 to 2 seconds) for complete slug-rod removal and the associated rapid-test-response recovery. Due to very high-permeability test conditions within the Hanford formation, formational test responses dissipated within the initial seconds (i.e., 90% recovery within ~3 seconds) of the test and are not discernable in the oscillatory pattern.

To provide a bounding, greater-than-K estimate for the Zone 1 tests, a series of over-damped slug-test type curves were generated for various high K values, using Zone 1 test parameters. Normalized plots of the type curves for K values ranging between 100 and 1000 m/day are shown in Figure 4.13. The plots in Figure 4.13 indicate a range of $K \ge 300$ m/day (and assuming $S_s = 1.0E-05$ m⁻¹) that correspond to rapid, formational test response recovery of 90% within ~3 seconds. This value range for the lower bounding limit for K is the best available estimate for the Zone 1 test/depth interval.


Figure 4.12. Example of Test-Configuration Induced Oscillatory Response for Zone 1, Well 399-4-14



Figure 4.13. Over-Damped Slug-Test Type-Curve Plots for Various High K Values With Zone 1 Test Parameters, Well 399-4-14

4.3.2 Zone 2 (Depth: 24.5 to 25.4 m)

After reaching a drill depth of 26.2 m bgs, the packer/well-screen assembly was lowered to a depth of 25.5 m bgs, and the 0.2985-m O.D. (11 ³/₄-in. O.D.) drill casing was retracted 0.8 m (i.e., from 25.3 to 24.5 m bgs), producing a test/depth interval for Zone 2 of 24.5 to 25.4 m bgs (bottom end-cap at 25.4 to 25.5 m bgs). The bottom of the open borehole between 25.5 and 26.2 m bgs collapsed after reaching the drill depth. The borehole geology log (Appendix B; Figure B.1) indicates that the test-interval section generally consists of sand, composed of >90% sand and <10% silt. The well-screen test interval was located ~11.5 m below the unconfined aquifer water-table surface, and the test/depth interval reflects sediments of the Ringold Formation (Unit 5).

While pumping during groundwater sampling (before conducting the projected slug tests on October 15, 2007), the entire well-screen section filled in with slough (sand) to a depth of 22.4 m bgs just above the top of the screen. Since "heaving" sand was an observed condition at this test/depth interval during drilling and borehole cleanout, slug tests were not performed and no slug-test results are available for characterizing hydraulic properties for Zone 2.

4.3.3 Zone 3 (Depth: 31.9 to 32.7 m)

After driving the 0.2985-m O.D. (11 ³/₄-in. O.D.) drill casing to a depth of 31.9 m bgs and then reaching a drill depth of 35.1 m bgs, the bottom 1.1 m of the borehole filled in with sediment slough. The packer/well-screen assembly was lowered to the bottom of the borehole at a depth of 34.0 m bgs, exposing the temporary screen to the formation at a depth interval of 31.9 to 33.9 m bgs (bottom end-cap at 33.9 to 34.0 m bgs). While pumping during groundwater sampling before conducting the slug tests, the bottom 1.2 m of the well-screen section filled in with sediment slough (i.e., fine-grained sediments), effectively reducing the test/depth interval for Zone 3 to 31.9 to 32.7 m bgs. The borehole geology log (Appendix B; Figure B.1) indicates that the test-interval section generally consists of a silty sandy gravel unit, composed of 50 to 75% gravel, 30 to 40% sand, and 15 to 20% silt. At the time of testing, the well-screen test interval was located ~18.7 m below the unconfined aquifer water-table surface, and test results reflect sediments of the Ringold Formation (Unit 5).

A series of three slug-injection tests and three slug-withdrawal tests (four low-stress and two highstress tests) were conducted between 1322 hours and 1511 hours (PST), October 19, 2007. The slug tests were conducted with two different-sized slugging rods, one with a volume of 0.0055 m³ and a larger one with a volume of 0.011 m³. These slug-rod volumes imparted a theoretical applied stress level of 0.68 m for the low-stress tests and 1.36 m for the high-stress tests within the 0.1016-m (4-in.) I.D. temporary casing string. Downhole test-interval response pressures during testing were monitored with a 0- to 69-kPa (0- to 10-psig) pressure transducer set at a depth of ~16-m bgs. The static depth-to-water for the test interval measured before testing was 13.14 m bgs.

All slug tests exhibited over-damped (exponential-decay response) homogeneous formation behavior, which is indicative of low-to-moderate permeability test-zone conditions. A comparison of the normalized, higher and lower stress, slug-test responses indicated nearly identical behavior. Slug tests exhibiting this type of response behavior can be analyzed quantitatively with homogeneous formation analysis approaches, as described in Butler (1998). For the homogeneous formation analysis, the standard typecurve method provided estimates of K ranging between 2.20 to 2.85 m/day, averaging 2.48 m/day, and S_s ranging between 7.0E-4 and 1.6E-3 m⁻¹. A selected example of the test-analysis plots for this test/depth interval is shown in Figure 4.14.



Figure 4.14. Selected Type-Curve Analysis Plot for Zone 3, Well 399-4-14

4.3.4 Zone 4 (Depth: 36.6 to 39.1 m)

After driving the 0.2985-m O.D. (11 ³/₄-in. O.D.) drill casing to a depth of 36.6 m bgs and then drilling an open hole to a depth of 40.1 m bgs, the bottom 0.5 m of the borehole filled in with sediment slough. The packer/well-screen assembly was lowered to the bottom of the borehole at a depth of 39.6 m bgs, exposing the temporary well-screen to the formation at a depth interval of 36.6 to 39.5 m bgs (bottom end-cap at 39.5 to 39.6 m bgs). While pumping during groundwater sampling before conducting the slug tests, the bottom 0.4 m of the well-screen section filled in with sediment slough (i.e., fine-grained sediments), producing a test/depth interval for Zone 4 of 36.6 to 39.1 m bgs. The borehole geology log (Appendix B; Figure B.1) indicates that the test-interval section generally consists of a silty sandy gravel unit similar to the unit for the Zone 3 test interval (i.e., composed of 50 to 70% gravel, 30 to 40% sand, and 15 to 20% silt). At the time of testing, the well-screen test interval was located ~23.6 m below the unconfined aquifer water-table surface, and test results reflect sediments of the Ringold Formation (Unit 5) just above the Ringold Lower Mud unit.

A series of three slug-injection tests and three slug-withdrawal tests (four low-stress and two highstress tests) were conducted between 1250 hours and 1431 hours (PST), October 23, 2007. The slug tests were conducted with two different-sized slugging rods, one with a volume of 0.0055 m³ and a larger one with a volume of 0.011 m³. These slug-rod volumes imparted a theoretical applied stress level of 0.68 m for the low-stress tests and 1.36 m for the high-stress tests within the 0.1016-m (4-in.) I.D. temporary casing string. Downhole test-interval response pressures during testing were monitored with a 0- to 69-kPa (0- to 10-psig) pressure transducer set at a depth of ~16 m bgs. The static depth-to-water for the test interval measured before testing was 13.01 m bgs.

Five of the six slug tests exhibited elastic, over-damped (exponential-decay response) homogeneous formation behavior, which is indicative of low-to-moderate permeability test-zone conditions. Data for a sixth slug test (slug injection test #3) could not be analyzed because of a perturbation in the test response at an elapsed time of ~100 seconds. A comparison of the normalized, higher and lower stress, slug-test responses for the five analyzable tests indicated nearly identical behavior. Slug tests exhibiting this type of response behavior can be analyzed quantitatively with homogeneous formation analysis approaches, as described in Butler (1998). For the homogeneous formation analysis, the standard type-curve method provided estimates of K ranging between 0.93 to 1.12 m/day, averaging 1.04 m/day, and S_s ranging between 1.0E-4 and 2.0E-4 m⁻¹. A selected example of the test analysis plots for this test/depth interval is shown in Figure 4.15.



Figure 4.15. Selected Type-Curve Analysis Plot for Zone 4, Well 399-4-14

5.0 Hydraulic Conductivity Depth Profile

Figure 5.1 through Figure 5.3 show depth profiles of the vertical distribution of hydraulic conductivity values determined from slug tests conducted at the 300 Area VOC wells 399-2-5, 399-3-22, and 399-4-14, respectively. The distributions are based on test/depth slug-test characterization results summarized in Table 4.1 and Table 4.2. The unconfined aquifer (not shown) in Figure 5.1 to Figure 5.3 lies between the water table and the top of the Ringold Lower Mud unit. As indicated, the limited vertical profile information suggests a decrease in hydraulic conductivity with depth at two of the three well site locations, 399-2-5 and 399-4-14. For well 399-3-22, the vertical profile indicates the highest hydraulic conductivity within the Hanford unit and lowest hydraulic conductivity within the upper section of the Ringold Formation (i.e., fine-grained unit). The hydraulic conductivity depth profiles indicate that K values estimated for the Hanford unit are at least two orders of magnitude greater than K values estimated for the Ringold Formation. The vertical hydraulic conductivity profiles for these three wells are consistent with the hydraulic conductivity profile for 300 Area VOC well 399-3-21 provided in Spane (2007).



Figure 5.1. Hydraulic Conductivity Depth Profile for Well 399-2-5



Figure 5.2. Hydraulic Conductivity Depth Profile for Well 399-3-22



Figure 5.3. Hydraulic Conductivity Depth Profile for Well 399-4-14 (Note: Other Ringold Formation Fine-Grained Units Not Shown)

6.0 Conclusions

Slug-test analysis results were obtained for 8 of 10 planned test/depth intervals during the drilling and borehole advancement of three 300 Area VOC wells: 399-2-5, 399-3-22, and 399-4-14. Of the eight successful series of tests conducted, sediment filled in the bottom portion of the well-screen sections during pumping for groundwater samples before three of these series of tests, resulting in a smaller test/depth interval. Two of the test/depth intervals planned for slug-test characterization could not be tested because of sediment filling the entire temporary well-screen section. Following completion of each of these wells, slug-test results were obtained for one test/depth interval within the final well-screen section.

Results from the 300 Area well slug tests provide general vertical distribution of hydraulic characterization information, for the upper, middle, and lower sections of the unconfined aquifer. The upper section of the unconfined aquifer lies within the Hanford formation (Unit 1), and the middle and lower sections occur within the Ringold Formation (Unit 5).

For test-depth intervals within the Hanford formation (Unit 1), slug-test responses dissipated within the initial seconds of the tests, indicating very-high-permeability conditions. These high-permeability conditions were confirmed by analyzing for a rapid, exponential-decay (over-damped) test response recovery of 90% within ~3 seconds. Analyses indicate a range of $K \ge 300$ m/day for two of the well sites (399-2-5 and 399-4-14) and a range of $K \ge 400$ m/day for the third well site (399-3-22). These hydraulic conductivity ranges were derived for test-interval sections that ranged from 0.5 to 1.1 m in length. These lower bounding hydraulic conductivity values are comparable to the general range of lower bounding values of >100 to >2,000 m/day and to the estimate of 568 m/day for other Hanford formation test/depth intervals recently tested in 300-Area characterization boreholes.

All test/depth intervals within the Ringold Formation exhibit exponential-decay (over-damped) slugtest response behavior. This type of slug-test response pattern is indicative of test intervals with low to medium permeability. Analysis of slug-test data for these test intervals indicate an average, test-interval hydraulic conductivity ranging from ≤ 0.01 to 2.48 m/day for the Ringold Formation (Unit 5). The hydraulic-conductivity estimates were derived for test-interval sections that ranged from 0.6 to 2.9 m in length. These average hydraulic conductivity values are comparable to the lower range of 0.04 to 41.2 m/day for 16 other Ringold Formation test/depth intervals recently obtained for test characterization boreholes in the 300 Area.

The limited hydraulic conductivity depth profile information for the three 300 Area VOC wells suggests a general decrease in hydraulic conductivity with depth. An exception is a slightly lower hydraulic conductivity in the upper section than in the lower section of the Ringold Formation at one of the well-site locations.

7.0 References

Bouwer H. 1989. "The Bouwer and Rice slug test – an update." Ground Water 27(3):304-309.

Bouwer H and RC Rice. 1976. "A slug test for determining hydraulic conductivity of unconfined aquifers with completely or partially penetrating wells." *Water Resources Research* 12(3):423-428.

Butler JJ, Jr. 1998. *The Design, Performance, and Analysis of Slug Tests*. Lewis Publishers, CRC Press, Boca Raton, Florida.

Butler JJ, Jr. and EJ Garnett. 2000. *Simple procedures for analysis of slug tests in formations of high hydraulic conductivity using spreadsheet and Scientific Graphics Software*. Open-file Report 2000-40, Kansas Geological Survey, Lawrence, Kansas.

Hyder Z and JJ Butler, Jr. 1995. "Slug tests in unconfined formations: An assessment of the Bouwer and Rice technique." *Ground Water* 33(1):16-22.

McElwee CD. 2001. "Application of a nonlinear slug test model." Ground Water 39(5):737-744.

McElwee CD and MA Zenner. 1998. "A nonlinear model for analysis of slug-test data." *Water Resources Research* 34(1):55-66.

Reilly TE, OL Franke, and GD Bennett. 1987. *The principle of superposition and its application in ground-water hydraulics*. U.S. Geological Survey, Techniques of Water-Resources Investigations, Book 3 Applications of Hydraulics, Chapter B6, Reston, Virginia.

Spane FA, Jr. 1993. *Selected hydraulic test analysis techniques for constant-rate discharge tests*. PNL-8539, Pacific Northwest National Laboratory, Richland, Washington.

Spane FA, Jr. 2007. 300-Area VOC Program Slug Test Characterization Results for Selected Test/Depth Intervals Conducted During the Drilling of Well 399-3-21. PNNL-16732, Pacific Northwest National Laboratory, Richland, Washington.

Spane FA, Jr, PD Thorne, and DR Newcomer. 2001a. *Results of Detailed Hydrologic Characterization Tests – FY 1999.* PNNL-13378, Pacific Northwest National Laboratory, Richland, Washington.

Spane FA, Jr, PD Thorne, and DR Newcomer. 2001b. *Results of Detailed Hydrologic Characterization Tests – FY 2000.* PNNL-13514, Pacific Northwest National Laboratory, Richland, Washington.

Spane FA, Jr, PD Thorne, and DR Newcomer. 2002. *Results of Detailed Hydrologic Characterization Tests – FY 2001*. PNNL-14113, Pacific Northwest National Laboratory, Richland, Washington.

Spane FA, Jr, PD Thorne, and DR Newcomer. 2003a. *Results of Detailed Hydrologic Characterization Tests – FY 2002*. PNNL-14186, Pacific Northwest National Laboratory, Richland, Washington.

Spane FA, Jr, JJ Butler, MD White, and TJ Gilmore. 2003b. *Improving Pulse/Slug interference tests for contaminated site hydraulic property characterization*. SERDP Fiscal Year 2004 Research Proposal, submitted February 25, 2003.

Spane FA, Jr and DR Newcomer. 2004. *Results of Detailed Hydrologic Characterization Tests – FY 2003*. PNNL-14804, Pacific Northwest National Laboratory, Richland, Washington.

Spane FA, Jr and DR Newcomer. 2008. *Results of Detailed Hydrologic Characterization Tests – Fiscal and Calendar Year 2005.* In Press, Pacific Northwest National Laboratory, Richland, Washington.

Springer RK and LW Gelhar. 1991. "Characterization of large-scale aquifer heterogeneity in glacial outwash by analysis of slug tests with oscillatory response, Cape Cod, Massachusetts." In: *U.S. Geological Survey Water Resources Investigations*. Report 91-4034:36-40.

Thorne PD, MA Chamness, FA Spane, Jr, VR Vermeul, and WD Webber. 1993. *Three-Dimensional Conceptual Model for the Hanford Site Unconfined Aquifer System, FY 93 Status Report*. PNL-8971, Pacific Northwest Laboratory, Richland, Washington.

Williams BA, MJ Nimmons, RE Peterson, BN Bjornstad, DC Lanigan, RJ Serne, FA Spane, and ML Rockhold. 2007. *Limited Field Investigation Report for Uranium Contamination in the 300 Area, 300-FF-5 Operable Unit, Hanford Site, Washington*. PNNL-16435. Pacific Northwest National Laboratory, Richland, Washington.

Zurbuchen BR, VA Zlotnik, and JJ Butler, Jr. 2002. "Dynamic interpretation of slug tests in highly permeable aquifers." *Water Resources Research*, 38(3):10.1029/2001WR000354.

Appendix A

Slug Test Field Notes for Wells 399-2-5, 399-3-22, and 399-4-14

Well 399-2-5, Zone 1



Page 2 of 2 PNNL SLUG TEST FIELD MEASUREMENTS - DURING DRILLING (continuation sheet) Test Date/Time: 9-7-07 Well ID: C5708 Borehole ID: 379-2-5 Test/Depth Interval: 33.3 - 35.9 ft Field Notes: Time: (set nod on bottom Large slug rod lowered to 1.0' above bottom of well and then we asure 1.0') Slug withdramal test #2 (very little test response) Dt=0.255ec. 0850 0855.30 0857 Lover vol to partially submered level. 0902:40 Slug withdrawal test #3 (large nod partially submarged) K 3.4' Submered Very little response. 0906 Lover large rod back into water column. No sense in conducting slug test m/ small nod because of no sesponse, so we'll conduct one more test using the large slug rod. 0910 Slug nitudranal test #4. Little response DRM Large slug rod partially submerged for tests #2, all tosts 2,3,4 1' off bottom of well, i.e. 3.4' of rod Submerged. 0913 logsproff Dom loaded data to file CRIOX_X16621_C5708_33-36ft. dat 0914 Large slugrod has a stamp "0.390" ft3 Prepared by: DRh Darrell New coner Date: 9-7-07 Reviewed by: North Rob D. Machly Date: 1/9/08 2007/DCL/SlugTest/002 (04/04)

Page _2_ of _3_ FIELD ACTIVITY REPORT 5 **Drawing Continuation Page** Date: 9-6-07 Well ID: CS708 Well Name: 399-2-5 Location: 399-B 300-FF-5 OU Continuation of Report No.: 3 water sample 1325 DTW= 37.4' 1+0C (9/6) (31.7'bas) 9/6/07 6.6 - Purged 23 min. @"10.7 gpm:"246gd Ground surface - Pump failed prior to sampling, sample will be collected on 9/7. - when removing the pump, the pump hitched up at the sub & pulled the screen up 0.4'. Bottom scilen is @ 36.3'ms 9/7/07 - Tripped Immp in, too large to fit, decided to start slug testing & remove screen setore sampling. - Prior to collecting weter sample, 4 withdownal fests Packer ne installed were performed, the server was removed, the caring was advanced to 35.5' by s 0.52'sub O.1' blank # the bare hole cleaned ow, to 35.0 bys (0.5' overlap. - 1041-1045 (4 min.) @ 3gpm= 12gd V DTW= 31.5 bas 1045 - 1115 (30 min) @ ~10gpm = 300 gel 33.3 - screen 6 sch-zo 30-slot (10.041) 12 + 300 = "312 galling 3.0 3.4 sample intr. I-13, BIP194 Blank/cap 36.745 400 . al. (0.37) 36.3' bgs Reported By: J. Horner Reviewed By: Title: Groologist Date: 9-6-07 Title: Date: Honce Signature: Signature: A-6003-711 (06/03)

Well 399-2-5, Zone 2

Page _____ of _____



Page 2 of 2

PNNL SLUG TEST FIELD MEASUREMENTS - DURING DRILLING (continuation sheet) 9-19-07 Test Date/Time: ___ Borehole ID: ______5 25708 Test/Depth Interval: _____70.0-73.0 _____ft bgs Time: Field Notes: Slug injection test #1 : 2-3 see to completely submare 2"00 rod 1631 into water column. At= 0.5 sec Slug inthdramal test #2 ~1 sec to pull not out of mater column, much 1649:36 quicker than injecting the slug rod. st=0.55ec. 5-6 min to fully recover Remove 2" DD rod from well and replace it with a 3" UD rod. 1103 Change at to Isee 1102 Loner slug nod into water column. Test #3. Drillers must go to lunch now and 1108:51 they won't be here to regulate pocker pressure. Genbasist will regulate pressure, so now test won't be disturbed. Turn compressor off 1122 Notired that packer deflated - shows up on water-level plot. 1127 Inflate Dacker and allow water levels to stabilize 1220 3" UD rod was fully submerced at= 1 sec 1234:30 Slug withdranal test # 4; 3"OD rod good test 125930 Slugification test # 5 3" O.D. rod was fully submarged Slug withdranal tost #6 1319_ Switch slug rods -back to singler one (i.e., 2"UD) 1334 Slug injection test #7; 2"OD slug and fully submerged 1339 Slug withdramal test #8: 2"00 R 1352 1407 Turn lagge off Dan bad date to file name CRIOX_X16621_399-2-5_7073fr. dat Packer test, pay Scal between 4" and outer casim 1414 1414:40 Then pump off to pour Seal D/W = 41.5' btoc (suter cosing) 1420___ D/B = 81.3' btoz (outer casmy) 1421 Note: Jake Horner (geolaist) has a nice diagram of packer/ soren assanly Danelf Prepared by: Newsmely Dawell Newsonthe Date: 9-19-07 Reviewed by: Reputs RobD. Machen Date: 1 4/08

^{2007/}DCL/SlugTest/002 (04/04)

	ACTIVITY REPOIng Continuation Pag		Page _ of _ 3
	sontinuation Pag		Date: 9-19-07
Vell Name: 399-2-5		Well 1D: 25708	
ocation: 300-FE-5 QU		Continuation of Report No.: V	
S.5 π^{4} 3.5 π^{4} 3.5	$\begin{array}{c} P \\ G \\ S \\ S$	19-07 Pumped w.s. /slu imp on: 0650° Intake ample time: 0800 (Main both imp off: 0855° ge @ 70 min. @~6 gp ge @ 115 min. @~6 gp % drawdown (packer Mot ime PH Temper Con 25 6.49 15.9 26 135 6.745 15.7 26 40 6.868 16.2 26 50 7.201 15.8 26	f = 420 gallons $m = 420 gallons$ $m = 690 ga$
eported By: J. Horner		Reviewed By:	
tle: Geologist	Date: 9-19-07-	- Title:	Data
	41107		Date:
gnature:			

Well 399-2-5, Zone 3

Page____of___Z



Page 2_ of 2

PNNL SLUG TEST FIELD MEASUREMENTS - DURING DRILLING (continuation sheet)

	Continuation sheet)
Test Date/	/Time:9-27-07 to 9-28-57 Well ID:65708 399-2-5
Test/Depth	h Interval: $\frac{16.6 - 12}{100}$ ft 2.2007 ft 2.2007 ft
-	123 126 (125) 1/4
Time:	riela Notes: 0709
1527	The sing sing sing the bond work small
	- slug rod.
	Compressor must be twined off before everyone leaves to day so we'll
	(maket sing withdrang) test tomorrow morrow
	Reset at from Isee to zuse and monitor over night
0645	Driller and spolgist provided breakfast! 9-28-57
0719	Download data
0735	Plot shows that water level is still recovering
0851	After consulting with Frank Spare, we will influte the packer system
0853	Water level is rising in response to packer inflation.
0902	Water level besimping to stabilize at 9.22'
0930	China at E 3000 1500 your non sot it to work - raining out . lyge met
932	Change at from 30592 to 5582.
	Have no watch - lost it the other day, so we'll use the logger click
	tomitiste Shy nitrational and injection (yelizal toil, 30min each, 20
3943	hitndravat Slug rad. (2" ral)
1013	Inject Sly rol (2")
1216	Data shows some uscillations - checked packer pressure and it is ak at
	Noizy data
1043	Slug mithdranal (2" rod)
113	1/2 cycles of slug testing traished.
1116	Driller's trosh puny for pound 5 gol of water cloesn't work, so we can't
	conduct packer feit
1117	Remove sel from well and get D/W + D/B
118	D/W= 38.25' btor (outer (asin)
120	D/13= 130.0' bloc, so (30.0'-5.4'= 124.6' by 5.7
	Turn pucker of f (deflak it) / Stasling
	Davalad data to file CRUN VILLEL 200 2 / 10 CLUL LLE LLE
	where D/B is because
	1 of silt on butting
spared by:	An Davel Wenning Date: 9/24/57 Reviewed by Chip Mark Cub Markley and Under

2007/DCL/SlugTest/002 (04/04)

Well 399-2-5, Final, Completed Well-Screen Section



Page 2 of 2

PNNL SLUG TEST FIELD MEASUREMENTS - DURING DRILLING (continuation sheet) Test Date/Time: _____12/3/07 Borehole ID: ______ C5708 260'- 41.0' ++ bgs (final screen) Test/Depth Interval 3237 spin Field Notes: Time: Slug withdrawal test #1; small rod - very little response 0816.3 0818 Slug injection first #2, 3 mail and - not much fest response Change to large slug rod 0922:30 Sluginjection fast #3, large rod 0524:30 Slug with drawal test # 4, large rod Stow million - bad test 082630 Slug million test #5 0828 Slug withdrame 1 test #6 ... 05417.12 Slugingertion fest #7; 1 age and - let endle you slock and draged not very gaitkly. 2850.50 Slug injection first 45; large of - let call go slock and dropped large and way quickly -wearly instantions sling displacement Dumbal data and save it to file CRNX. X16621_399-2-5=28-41H. dat Davrell Neudone Dote: 12/5/07 Reviewed by (100144 Rib Mail 16 Dor Prepared by:

2007/b0./SkgTest/002 (04/04)

Well 399-3-22, Zone 1

Page_____of_____



Page ____ of ____

PNNL SLUG TEST FIELD MEASUREA	NENTS - DURING DRILLING (continuation sheet)
Test Date/Time:	Well ID:
Test Date/Time:55.5-55ft 653	Borehole ID: <u>25708</u>
Time: Field Notes:	
USA230 _ plug injection test # 5,	large stug of, sscilla Fon, response
A Hacked XD ~2.5' bs	low long sing rod
0315 Lowerld large and comple	tily Submered in waterstromm.
0821.30 5/49 with 45+#6; lang	6 sing red with XD attached to red.
JSall 5 Akn	
US26 Lower rod into water column	
- Charles Charles Handler	in the line of the last the set
0827.32 5145 with tist # 7; 16xc	stres vod wrth XI) citizi han to ion
Change back to small	
0836 Lower XD to 1-2 ft ,	nto water brown.
0041 Sling the China 1131 43 Sh	noll slue, road; sse, llatory test response
2343:30 - 3145 KITA. 1107 #1.	443121
$\frac{0343.05}{0853} = \frac{3112}{0100} \frac{1111.00}{0100} \frac{147.3}{000} 147.$	'') = 60.5 = 55' bis
0337 0160 0102 ()	
Data doubt which to file	CRIVX-X16621-399-3-22-53-55ft. dat
Prepared by: L. R.M. Dawell New Sond Date: 1	16/37 Reviewed by: 10/10 Kib V. Machagare: 1909
angu prou	2007/DCL/Slug Test/002 (04/04)

Well 399-3-22, Zone 2

Page_/ of 2



Page 2 of 2

PNNL SLUG TEST FIELD MEASUREMENTS - DURING DRILLING (continuation sheet)

me: Field Notes: 453 $7.88'$ 456 $5lug with 528 7.89' 533 D/R 539 5lug 626 9.92' 528 10.470' 528 10.470' 528 10.470' 528 10.470' 528 10.470' 5658 10.470' 777 D/W 7730 Packer $	$\frac{25-85.6}{7.5-79.7} \frac{300}{51} \frac{1}{55} \frac{1}{$	Borehole ID: <u>Slug rod (0.190 ft3)</u>	C 5-708	
me: Field Notes: 453 $7.88'$ 456 $5lug with 528 7.89' 533 D/R 539 5lug 626 9.92' 528 10.470' 528 10.470' 528 10.470' 528 10.470' 528 10.470' 5658 10.470' 777 D/W 7730 Packer $	thdrawal test #2; small			
453 7.88 456 Slug un 528 9.89 533 D/R 539 Slug 539 Slug 626 9.92 524 Set d 5658 10.476 777 D/W Try to 730 Packer		slug rod (0.190 ft3)	At=1sec	
$\begin{array}{c c} 456 & Slug with \\ 528 & 9.89 \\ 533 & D/R \\ 539 & Slug \\ 539 & Slug \\ 626 & 9.92^{1} \\ 5658 & 10.476^{1} \\ 5658 & 10.476^{1} \\ 7717 & D/W \\ 7717 & D/W \\ 7730 & Packer \\ 7730 & Packer \\ 7760 \\ 7710 & Packer \\ 7710 &$		slug rod (0.190 ft3)	nt=1sec	
$\frac{528}{539} = \frac{9.89}{5105} \frac{9.89}{5105} = \frac{9.89}{5105} \frac{9.89}{5105} \frac{9.89}{5105} = \frac{9.92}{500} \frac{9.92}{500} = \frac{9.92}{500}$		slug rod (0.190 ft3)	At=1sec	
	= 82.2(1+2)(1+2) =	-		
539 <u>Slug</u> <u>Veg</u> 626 <u>9.92'</u> <u>Set a</u> <u>5658 iu.476'</u> 7717 <u>D/W</u> <u>Try to</u> 1730 <u>Packer</u>	= X2.2' h h u (u'') =			q2
Veny 626 9.92' 5et s 5658 10.476' 7717 D/W Try to 730 Packer				2.
5et s 5658 10.476' 7717 D/W Try to 730 Packer	insection test #3, large si	lug rod (0.395f+3)		
5et s 5658 10.476' 7717 D/W Try to 730 Packer	slow test response			
0658 10.476' 1717 D/W Try to 1730 Packer				
1717 <u>D/W</u> <u>Try to</u> 1730 Packer	t to Ionin overnight	and the second		
1717 <u>D/W</u> <u>Try to</u> 1730 Packer			11/7/07	triday
1730 Packer	Downloaded data to 1.	aptop computer	- 1	
730 Packer	= 46.38' bt. (4")		bys	2
	Iswer tonnoduce past si	us isd		¥3 *
231 Set to				
	insuluciar at a deeper fere,	18.09' 51	us al was pulled	dont.
	atér column to fræ tran			
737:10 Slug,	his chan test #4 - lang	65/45 rok		
tit Not re	3= 79.9' btsc (4")	lon test		
<u>244</u> D/I	3= 79.9' btsc (4")	Silted to topot se	incon	
	(Kiox-XI	6621-		
Data	dambaded to file 39	19-3-22-77-79ft. dat		
			• • • • • • • • • • • • • • • • • • • •	
			P./14.8	
pared by:				

^{2007/}DCL/SlugTest/002 (04/04)

Well 399-3-22, Zone 3

(no field notes—slug testing abandoned)

Well 399-3-22, Zone 4 (Final, Completed Well-Screen Section)

Page _/ of 2



2007/DCL/SlugTest/001 (05/04)

Page _2 of _2

Test Date/Time: _____12/3/07
 Test Date/Time:
 12/3/07
 Well ID:
 399-3-22

 Test/Depth Interval:
 125.2 - 134.7 ft b5s (final screen)
 Borehole ID:
 C 57086 Arr
 Time: Field Notes: 1407 9.0968' 1416 9.68821 1419 Initiate slug withdrawal test # 2 using small-dia rod Sluw recovery, over-damped test response. 1444:30 Initide slug injection test #3 longe nod over-damped test response ~20 min to fully recover 1915 Initiate slug withdraw (test 4) large rod 1352:30 Initiates slug injection test #5, small slug rod 15:2 Initiate Slug withdrawal test #6, small slug rod 1532 D/W=4662' bloc (arter protective casing top) Download data and save to file CRIOX-X16621-399-3-22_125-135,00t Daniel Marina Prepared by: DRM Dawell New const Date: 12/3/07 Reviewed by: DWM _ Robe D. Mackleyhte: 1/1/00 2007/DCL/SlugTest/002 (04/04)

PNNL SLUG TEST FIELD MEASUREMENTS - DURING DRILLING (continuation sheet)

Well 399-4-14, Zone 1

Page ____ of ____



Page ____ of _2

PNNL SLUG TEST FIELD MEASUREMENTS - DURING DRILLING (continuation sheet)

	PNNL SLUG TEST FIELD MEASUREMENTS - DURING URILLING (continuation sheet)
Test Date/T	ime:10/10/07 Well ID:399-4-14
Test/Depth :	Interval: 476 555.1 ft 655 Borehole ID: C5707
	ine:
Time:	Field Notes:
1041	Red moved up a little to get it off bottom of well.
1043	Sluguith. #est #2 using small od; Small oscillatory response this hure.
	Leave we and the coll and level (and a test)
1045	Lower small slug vol to fully submerged level. (not a test).
1048:30	Slug with. test #3, small slug rod. (0.195 ft3)
1105	Remove small slug and and attach large slug and to cable
	Lower sly rock slowly into mater column volume of rock is (0.390 ft3)
	Slug with test # 4; large rod raised very quitk 4 (21500)
1/11:30	Shiginjection test #5 (lowered tod in 1-2 sec)
1114	Slug nithdrawal test #6 (rod railed in ~1slc)
	Uscillatory responses look good on realtime plots for tests # 3-#6
	water .
1116:15	Slugingertion test #7 (1-2 sec to lover roch all the way in column)
	- Arel Slug rod
1119:30	
	4" carsing with sky sol. Rad is jammed - Driller will try to tap
	on topot 4" casim to vibrate vod and transduce losse. Doesn't work
150	Driller got transducer unstuck by pulling with a third line. Transduce
	does not appear to be damaged. We will not conduct any more slug tests
	in this well.
	D/W= 47.7' btoz D/B= 56.6' btoc
	Saved downloaded data to file CRIOX_X16624_399-4-14_47-51ft. dat
Prepared by:	WRN Linell Neuranne Date: 10/10/27 Reviewed by: Reviewed by: R.D. Mackley Date: 19/00
	sign print sign print

Well 399-4-14, Zone 2

Page _____ of _____



Well 399-4-14, Zone 3

Page / of Z



Page _2 of _2

PNNL SLUG TEST FIELD MEASUREMENTS - DURING DRILLING (continuation sheet)

	Time: _10/19/07		Well ID: _1657	o7 <u> </u>
Test Date/	Interval: 104.5-107.2'ft		Borehole ID:	-4-14) alm
rest/Depth				
Time:	Field Notes:			
1436	Rig June back m - Slug withdrawl. Good	Sperna intuference	1 ~	2,000
438	Shia withdrawl. Good	Clean pull. Kig TI	and - inchit anter	······································
1440	fully recound in sta			
1455	Lowarn larger yod	le al hale A	hand had been at an	He ilservel.
1505	Slue Miechen (~ 4	2000 1010 TH 7)	1 TOTAL AND THE OFFICE	
1520	fully recovered high			
1522	Withdrawl of large			
1529	fully recovered. (~4			
1531	Ria On, Pulling up	large Stur vod Sou	ne tangling but not r	us bad.
1539	DTB = 107.0 bas	: 10,2' silted	in the Scipen. Wa	er level
	has visen about 0.25	Since first ant x due	er m hole (How P 9	. 81 of press)
1554	Slug mighting in I sma	11 rod. (-2.5' 1051	pouse). Clean push;	no tanale.
1603	fully recovered. Collect She withfrow 1 2.14	a static prose ins.	Very stable.	
1611	She withdrawl2.14	cespanse.		
1619	tuth recovered			
1631	Pulling small rod out	+ hole		h. f. 1. 140
1636	Pourd 5 aul 4.0 W.LSY2.8 bas; DTI	Allow Annulse MC	lesponse = packer	st laking
1637		S = 106.9, 10TO >	uting of ECAS thomas	2 m + 18 201.
1645	LPOUR Sit.	L'EDD Sim	Test Data vom	ji) in 2 m Z vic
	<u></u>	e: (5701-310g	1+31 Sula Juna	10-11-001-115
			· · · · · · · · · · · · · · · · · · ·	
			at	
			· .	
Prepared by:	RD WOLKING ADDING	te: 1115/07 Reviewed by:_	DRN Damell Num	Contra 1/8/08
	sign i print		- <u>-</u>	'SlugTest/002 (04/04)

Well 399-4-14, Zone 4

Page_1_of_3_



Page 2 of 3

	PNNL SLUG	TEST FIELD M	EASUREMENTS -	DURING DRILLI	NG (continuat	tion sheet)
Test Data /Tim	me: 10/2	3/07			Well ID: _	399-4-14
Test Date/ In	nterval: 120	-128.2	ft bas		Borehole ID: _	C5707
		L NOTE: Test	tinternal cha	med slightly a	fterscreen	silted in.
	Field Notes:					
1349	Driller	made mark	Sunceble	0	80.03)	full of the off
1350	Slug inj	iction test #1	using small	slug rod (0	<u>, (175 FT-)</u>	fully submerged
	5/04	ecoving ~	10 min tor tull	Of Line	Jund Cxps	contral decay curve.
1407:30	Digni	holocural test	I HE SMALL Y	ad tully moment	a	
		NO min fu	" Fectorery	11. Pisturbe	dxp	
1424	Kemore	121 2'	bloc outer c			
1960	0/0	1771 h	at 1 cane	n cited in	0,5 Since	e slug tests started.
	B	Conduct or	re injection a	nd one witho	Iranal tes	+ using large slug (0,390)
	rod t	minimize	more silting	in of 6"I.	D. Screen	
1433:30	Silver	icetion test	·#3. lane	slug rod (C	370++3)	tully submered
1435	Pertu	-bation in S	lus test respon	nce at te=	IS min.	quicker recovery at t>1.5
	Full	recovery ta	kes ~10 min.	$H_{6} \approx 0$	4 f4	
		,		r= 2"		
		0.390 ft3 =	$=h\pi r^2 = h$	(2/12) ² =	h (0.08	$722 fb^2)$
	<u></u>	h= 0	0. 390 H	4.471		
		0	0012211	0	Clubber	. 0
1451	Slugh	ithdranal	+est #4;1	large slug vod	tury submer	to lest
15.0	- Over-	<u>tamped, expo</u>	nential decay	cure (no	04-T-PDatis	n); good test
1507	Kemer	2 1214'1	to ater	Enell.	= 12	7.8' b35; 4" hasn't
1312	<i>1</i>	12 d c.		end of test #	2	/.0
1514	Slue	iection tes	st # 5. sme	ill clus sol	· 3000l t	est, overdanged response.
12.1		geographic -	Hore = 2	24' inside	4" I.D. Cas	my and 0,195ft volume.
1531	Sluga	thdrawal t	test #6; sma	all slug rod fi	Ily submary	ed
	-		test respone	0 · · ·		
1545:10	Packer	integrity tes	t - besen por	un Sgal do	un annul	us between inne
	and	uter casing		0		
1545.45	End	suny !	No response	-looks soal.		
Prepared by: _	WR.H sign	Danel 1 News	Date: 10/23/07	Reviewed by:	sign	<u>cb1). MacHeysate: 1/9/08</u>
	-					2007/DCL/5lug Test/002 (04/04)

	PNNL SLUG TEST FIELD MEASUREMENT	S - DURING DRILLING (continuation sheet)
at Data /1	Time: 10/23/07	Well ID:
st Date/ 1	Tratamel: 120-128.2 ft b55	
st/Deptn	2 NOTE: Test interval a	Borehole ID: C.5707
me:	Field Notes:	
1548	Turn logger off	
1550	D/B= 131.0' bloc (outer	(asing); so D/B= 131.5-3.6= 127.4 695
1552	D/W= 46.5' btoc "	(asing); SO D/B= 131.0-3.6= 127.4 bys '', SO D/W= 46.5-3.6= 42.9 bys CROX_X16621_399-4-14_120-128ft.dat
	Dannload data to fil	e CRIX_X16621_377-4-14_12572874.dat
		/
		/
	·	

2007/DCL/SlugTest/002 (04/04)

Well 399-4-14, Final, Completed Well-Screen Section

Poor 1 of 2 PNNL SLUG TEST FIELD MEASUREMENTS - DURING DRILLING 399-4-14 Test Date/Time: 12/5/07 Well ID: Test/Depth Interval: 42.8 - 58.0 ft bys (final well screen) C 5705 Borehole ID: Transducer S/N: 2437003 42.8 ftbss. Pre-Test Depth-to-Water: 2.3268 42.8 Multiplier: ft bis Post-Test Depth-to-Water: X16621 Logger S/N: ____ Measured Test Lengths A = Reference Pt (RP) to Top of Inner Casing F = RP to Bottom of Borehole (Pre- & Post-Test) B = RP to Top of Outer Casing G = RP to Top of Packer H + Packer Length 2.0 C = RP to Bottom of Outer Casing ft I = Screen Length 3.0 D = RP to Top of Screen E = RP to Bottom of Screen 8"SEH to potective . Reference Point = Ground Surface - casing νA ft Test Stress Information 2.0 11 MA last-Test 38.0 Test # Stress Applie E (2) 032F D 58.03 34 2 3 4 14 Inner Diameter (inch) NΑ ft 14 well-screen/ 6" drill casing _____NA 20.05 ft barehole <u>NA</u> Т lcap: 58.03'-60.53' 455 2 drilled to 136' bys bRh Danell Heuroman Date. Prepared by Rob D. Walder Field Notes: Time: 1/B = 63.7 Hoz (8" outer carsing) 1358 D/W= 45.8' Hoz = 45.8-3.0 = 42.8' bys 1359 Lugger clock is synchronized within 1 sec of wortch 1408 large glug rook : not much test response. 1418 Sher rejection test #1 betw. 2 and 3 ft Transduce as installed below water love nd 1421 Slue with drawal tost #2; large slue ٠. (Slow immersion) 1.1 142430 Slug injection K1 H3: (Slow with drowal) She withdranal tost #4.4: " ... 1427 annitable for the's 6" I.D. well. The 4"0.D. 4" op slug not is not 2007/bCL/SlugTes1/001 (05/04) red that Durotek has is being used.
Page 2 of 2

	PNNL SLUG TEST FIELD MEASURE	WENTS - DURING DRIL	LING (continuation	on sheet)
Test Date/1	ime:12/5/57 Intervel:42.8'~58.0'ft 655		Well ID:	397-4-14
Test/Depth	Interval: <u>42.8'~58.0'</u> ft 655		Borehole ID:	(5705
Time;	Field Notes:			
143028	Slug injection test #5; lan and very guizkly.	Ksluy todj let l	me go slack o	Idopped
1432	Slug mitudramel test #6	; lance stong roof	inst much .	lest response.
	Dismilial data and save -	& file CRIOX_>	166212399-4	14-43-58 A.dat
		/		
		/		
		/		
	/	/		
	/			
	/			
	1			
	L			
Prepared by: _	DRU Dameli Neucon Done: 12/3	Z Reviewed by:	son Robi	Machany Date 1/9/00

2007/DGL/SkgTest/002 (04/04)

Appendix B

Borehole Logs for Wells 399-2-5, 399-3-22, and 399-4-14

		BOREHOLE LOG				ge / of 4
: 05708	10/	ell Name: 399-2-5	Location: 3		Dat	
					Opera b	a Unit_
	ICE	Cheracterization		asuring Point:	Ground.	Surface
Sample	Graphic		Description			mments
Type Blows No. Recovery	Log	Group Name, Grain Size D Color, Moisture Content, So Max Particle Siz	istribution, Soil orting, Angularit e. Reaction to F	Classification, y, Mineralogy, ICI	Depth of Casil Method of Driv Sampler Si	ng, Drilling Method, ving Sampling Tool, ize, Water Level
	Vx XX	0 -13: Sandy (4	revel bac	kfill	Coble to	of drilling.
		Poorly sonted with	60-70	To F-VC	with de	ive barkel.
	5300	angula felsic-	dominat	& sand	advancin	4 113/4" Las
	2n	and 25-3570	pebbles,	cobbles	0	
1-2-07		and baylders (60	ulder pre	sent from	Geologic	archive samp
100 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	\mathcal{U}_{0}	0.1 40 0. 6 bg:). No ex	n poith HC	at approv	x. 5 intr. (A.S
	\mathcal{O}	5 5% silt pre	sent (mil	non). Soil	(A.S. Samp	les at: 5,10,
	30	13 olive brown	(2.57, -14	2	15, 20	0
	0	"12' -13' C	Genel	(.6)	Split speen	from 4.5' to
	DQ.	Poorly sorted st	Graver	(SG)	HEISE BI	lough 4.5-50)
AUT.	20	many sorrea, St. 1	SU WAR	3/2 : 1/2	1.6.1 2 0	PL45 (1606)005
	2000	40 40 809 SHU-	Ly fine	to is come	r oral + s	ollected e 4.5-7
		pebbles & cobble	2 > 70%	hard f)	lanct - S.S	chemout).
1 Te	000	No rxn with HCI	1 < 2070	m- vc sab.	Moisture t	in sumple@
9-5-07	600	angular sand (bese	It-domina	ked with	10' 6as (M.T.) BIPL2
See See	0.0	a bundant zones or	f oxidatio			
11-15	2000	-20' m.f sana	fraction	incuenses	split - Spoor	e 15-17 bas
5 66 5 6	3.0	m-f sand i	's generelly	y oxidized		IPLY6
	80	& felsic-d	minated	<u> </u>	1-Gel \$ 5	-G.l. G.S. @ 15-
9.5-07		-23 - 42 : SA	udy grave	Grevel	M.T. Samp	le @ ~17'bas
33460 19	878	Med. sorted with	72070 5	esalt - dom.		n s.s. shoe).
		sub ang. to sub-row	and pepties	+ small	HEIS#B	17128
4.5.07	OF,	Cobbles ~ 1590 m-	ve any to	sub-eng.	Split-speon	20-22.5 405
100 100	2000	Sand (course sand	s becalt-d	and med.	HEIS # = E	
4-12-100	122	sumse tine send	13 telerco	Com. [> 70%	M.S. @ 27	11
10 1 2 5	38	c-vc send 1 =	sto pale	of the (57, 94)	Ishoe of sp	6t-spon BIPL 29
20	2003	with Hel	= 10-15 c	1. Non	20'- 22'	S-Gal G.S.
1118 - 18	202	sanamild, year, las	<u>- 10-15 c</u>		Sul- and	23.5-27 %
1 1 1 1 1 3 B	Of St	- 30 matrix donne	sel ma	wel below	HEISH BI	PL48
1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	Offer	water table is	washed ph	ON With	M.T. 124	Sal builtot
	86	any a pepble	matrix.		samples à	3.5-26 bas
5-Gel stuffert	200				M.T. # BI	PL30
1-6-07	3. S				Split - spor	on 28-31.5
11 2 3 3 6 2 3	20				(slough 28	-30) BIPL49
	08	39'- coarse sand m	nature el	, neterolithic	MT. C 30	2' has (1/4 s.s.
Long T A	RE	cobbles to 2.5"	well-rown	Led.	#BIPESL	<u> </u>
° 1248	58 Co.	- sand is very a	tark (mafin	e)	5- gol but	ket spragle @
1[8	000		T		DTW= 31.8	bas (9-5-07)
ted By: JHorney	1		Reviewed By:	<u> </u>	Walker	
Gedogist		-	Title:	Geologis		
ure: bole Hen		Danali I -		and in		
ne. Jour form		Date: 9/6/07	- Signature:	P Wa	1kg	Date: 12-13-07

				BOREHOLE LOG		Page 2 of 4
						Date: 9-6-07
ell ID	CS	708	We	all Name: 399-2-5	Location: 300-FF-S	201
oject	TOP	- Cha	ractoriz		Reference Measuring Point:	
-		mple	ACT I/Z	.,	Description	Comments
pth		<u>.</u>	Graphic			
ï.)	Type No.	Blows Recovery	Log	Color, Moisture Content, S Max Particle Size	Distribution, Soil Classification, orting, Angularity, Mineralogy, ze, Reaction to HCI	Depth of Casing, Drilling Method, Method of Driving Sampling Tool, Sampler Size, Water Level
-			0000	40-42: sandy gravel	(SG) V. LOAVSE	Split Spoon 33.8'- 36.8'
-	2 2 2	Ino Ino	0000	matic sand me		160 Blows, 50% rec.
-	sgal quab igal guab splitspace	1 . 9:			bbles to 3" pro-4	HEISTBIPL50
-	sant sund splits prov	15 L	9820	sorted		1- Gali & S- Gal grabs
-		<u>ت</u> ا	o les	42-419' V. coonse son		33.8 - "36.8' bgs
-	1 and	كسماط	5	40% matic	dasts, bil gh dests,	
-	5901	1999			thic cobbles to 2",	Waster sample (pumped) from
-	5-3	5. 1%			hed, well-scated salt	35.5 bys to 27 bas (slowing
y-	- 1 - E			il a man pepper tex	-il l	present 35-37 6gs). No
-	STARS-	216 3		46.9-47.4 : mapsive		packer or script used.
-	engues sangues			1	udant clay, no coholes	HELS# BIPL94.
-	Plit					a) i - a- a' - a'
-		1	000	1 4 1 11		split sponn 40.e - 43.e
-	+ Salens			1	obbles to 3"	HEIS # BIPLSI
-	Sal	× .	0.0.	53-56 : sandy are	VC Contraction	
	- 4	6 8	0.0	sand (656)	devolithic cobbles to	igal t s gal grubs
1	-		ETT-		well-rounded, party	DTW = 33'(9-12.07)
-				sorted.	well - rounded, perry	510 - 33 (1.2.01)
-		11. S	11-1	56-605 marsive	silt unit (Ringold?)	split spoon 44.4 - 47.4"
-				meduin tan.	well - consolidated,	160 blows, 100 6 recovery
-				why plastic no l	KI reaction	HEIS # BIP LSZ
-		ų i		bois - 66 : morsin		Jen + bit se
-				color = d	ank green/green	pumped wroken sample 47-49
			1	46-69: meduin	group-green sound	HEIS #S BIPLAS, BIPLOS
				well-sorted	no cobbles, mostly	BIPL71, no packed or scie
		125.			o micas	
				- Il don't arou	(Gley 1 4/N)	split spoon 49.5-52.5'
						160 blows, 756 recover
				-72-73 fine	pepples present.	HEIS # BIPL53
		1.9 S.T.			/ / /	Punned water sample from
_				73' - 103' : San	dy Gravel (sG)	66-69 bas (stough 67-69'
_	. 1			Mod. sortal & v. d.	ask gray (Gley 3.5/2) with	HELS# BIPL97
-	1-18-07		. a	60% sub to well-	rounded vt-ve metic-	1-Gol grab @ 73-75' no
_	Split 3	*C.,	ϕ	dominated pebbles	1-40% m-c sub-ang.	enough recovery for 5-
_	BIPLSY	Ser.	0 °	felsic - dominated	sand (780% felser	grald sample.
_	246 6 600		00°°	mostly gtz + <20%	matre). Sund matrix	Pumped w.s. from 70'- 25
-			000 C	is the same as .	sund above, only	bas (succeed 70-73.4'; slo
-			0000	slightly more com	e. Some foortle	from 173.4' - 75' bas)
			\mathcal{O} oo	with sand heav.	ing near contact.	HEIS # BIPL97 (2-31).
porte	ed By: N	lichael	E. Ca	m J Horner	Reviewed By: L, D	Wolker
e:	5	- (1 .1	1 Gardages +	C (
	Sen	1 miles	alagent	Teologisi		
natu	Ire:	INC	100	u lan Date: 9-20-0	Signature:	Date: /2/13/0:

				BOREHOLE LOG		Page 3 of 4 Date: 9-20-07
Nell ID	: 657	08	W	ell Name: 399-2-5	Location: 300-FF-	
Project		Anen		Characterization	Reference Measuring Point:	
		mple			Description	Comments
Depth (Ft.)	Type No.	Blows Recovery	Graphic Log	Group Name, Grain Size D Color, Moisture Content, S Max Particle Size	Distribution, Soil Classification, orting, Angularity, Mineralogy, ze, Reaction to HCI	Depth of Casing, Drilling Method, Method of Driving Sampling Tool, Sampler Size, Water Level
0			00	73' - 103': Sa	1 1 1/1	Salit spnen 73-75'
_			00	- description on	ba#2	246 610WS, 100% rec.
-			00	AP +14-07	10	HETS# BIPL54
-			000	- 400 - Leovin	g sand, storded	GP
-			00	adding	water	W.S. +8-84'-87' bes
5—	6		00	103'-110': Sa	ed (5)	84.5'-87'LLC HEIS 510494
_	3.1		00	well-sorted me	dium Lelsic-dom	Greb Samples (G.S.)
			000	(60-70% felsic)	sub-ang sand.	G.S. @ 80' 85' 95'
_			000	·	0	100', 105', 110' \$ 1154
			00			
E.			000			
-			0.00			
-			000		·	
			080			
			000			
			00			
_			0000			
-			20			
2			00			
-			0.00		n ₁₁	
-			0,0			
-			RANK			
5 _	4.9					Rumped w.s. 105-110
>						screen to 108.7; slough
						108.7' to 110'
_						HEIS # BIPLIBO
_						
0			00.00	110-112 · Card.		
-			000	TIO TIP . Sahoy	Grand (sG)	
1			200	Mod. sort wit	h 50-75%	
-			000	med, - course sa	ad (~ 6090 Leber	
~ -			0000		wing me sand	
s—			000	+ sephles	160- 70% felsix	1
0.00			0.08	/		
			Lee-			
eport	ed By: -S	. Hown	ev	· · · · · · · · · · · · · · · · · · ·	Reviewed By: L.D.	Walker
	seolo				Title: Geologist	
ignatu	1	Le /L	men	Date: 9-26-0		Date: 12/13/07

				BOREHOLE LOG				Page _4 of _4 Date: 9-26-07
Vell ID	cs	708	We	ell Name: 399-2-5	Location:	300- FF	5 00	
Project:	30	O Ane	a. TCE	- Characterization	Reference N	leasuring Point:	Groun	1 - 0
Depth		mple	Graphic	0	escription			Comments
(Ft.)	Type No.	Blows Recovery	Log	Group Name, Grain Size Di Color, Moisture Content, So Max Particle Size	stribution, So rting, Angular	il Classification, rity, Mineralogy, HCI	Depth of Method of Samo	Casing, Drilling Method, f Driving Sampling Tool, ler Size, Water Level
20			0000	117'-120': Sil	y Sand	y Grovel	Archiv	/ /
_			0.00	increased si	77.		126-127	t 130'bas
-		W.5 5.T	000				Pumpe	L w.s. 123-127
5_			0-00			Parent Marker	123-	127 but tanged
-		2		125'-131: Silt	(m)	RLM	top or	e sedimente
				Gley (4,5G)	an greeni	sh gray	HEIS	# RIPLBI
				- 130' clay is	Jery da	ik growy		
so				(2.5y, 3.5/1)		/ /	Slug	test with
	1						a bove	for 42.5.
4					_		123-1	27, slough 125-12
5_				······································				
°_					a debata			
-								
_								
40						(0		
-								
-	1							
45_								
-								
50_								
-								
-								
is								
eporte	d By:	J. 40			Reviewed B	y: ().	Walke	
itle: /	Seolo	gist-	-	······································	Title:	Geologis	1	£
ignatu	re: 0	21.	L.	Date: 9-28-07		ne ul	16	Date: 12/13/07

				BOREHOLE LOG				Page of
								Date: 10-3/-07
/ell ID	05	706	We	ell Name: 349-3-22	Location:	300-FF-5	ou	
roject	1000		_ %	nvestigation_		leasuring Point:	Grown	d Surface
-,		mple			Description	······	4000	Comments
epth (Ft.)	Туре	Blows	Graphic Log	Group Name, Grain Size Color, Moisture Content, S Max Particle Si	(R)	I Classification,	Depth of	Casing, Drilling Method, of Driving Sampling Tool, oler Size, Water Level
	No.	Recovery	•	Max Particle Si	ize, Reaction to	HCI	Samp	ler Size, Water Level
<u>_</u>				0-0.5: Crush	a grave	from	Cable	tool drilling
-	e		ei = 50	old roa	d.	110	with	11 hu casing
			0	0.5-1.0: Silly 5	andy gra	vel (mol)	e dri	he barrel.
-				Poorly Sorted w/	6018 +.	to law	Antin	auch comeles Q'
			0.00	sand 13 Lolo	-7.07. 6.	14 range	1' 5'	grab samples @:
- ۲			9.90		2-/- 5/		23.5.24	5, 25 30 \$35'
2			00000	1.0- 2.0 : Silty	Sond (m S)	Moist	une tin (MT.)
1			000	well-sorted ma	derately a	consolid de	same	4 @ 9.5 bas
			000	+ light of brow	2n (2:54	, shy with	HEIS	# 131PL 33
<i>。</i>	A.T.		000	-7570 verm fels	ic-dom. s.	and + -25%		
_			000	silt. No rxn.	with HC	1. sl. moist.	M.T. 6	2 15 bys, 131PL3
_		1	50°					<i>v</i> ,
2-		ž	000	2.0-16: S. 14 S	andy Grays	el (mstr)	L	
-	-	2	00	Horly souled & It. y	ellowish .	brown(254,64		
5 —	A.T.		2()	with 30-+070	sab-ang.	popolise		· · · · · · · · · · · · · · · · · · ·
-		Ť	SKE AS	COODUL (3 +07, 60	salf Va	Blue Lan		
375			OF 33	te ang. Sana (tin	10% h.	It) with	M.T. 6	ad has BIPLS
		d	0.28	2209 silt N/2	in with h	HI max	a G	
	M.T.	ิ โ	0000	capples is ~25	Em. Une	nsolidated.	Bould	4 from 14-16 695
10-		3	878 B	· ~6' color chan	ues to non	(2.54, 5.5/1)		
		2	6120	16 -48: A': Silly	sandy Go	ovel (us G)	4 11-1-0	4
		15-	200	Poorly sugar in	ell-consol	ida Rd with		
	M.T.	4		70-80% basalt-	dom. pel	bles/cobble	M.T. G	2 24 the S BIPLS
Te				10-15,90 F-VL 4	rng. bass	lt-dom.		0,
<u></u>			220	sona \$ 1075	76 H. 121	lowish brow	r	
			9990	(2.5%, 6/4) sl. M.	ist silt.	Topot		
202			60 S P	Unit (@ 16 bgs)	has a sy	ring exp		
-			00	with HCL, whi	ich decu	all's to	47.	2 2 2 1 / 2 2 - 1
50 -	MT.		38	Ala to weak a	- 4 400	(.()	M.T. (
-		1.5	a co a s	Back carles	19vover	a de	- 11-1-0	/T
	1	ĸ	000	12 50 4/ Lance	~609- ~ 20	& are grey	M.T. 6	35 14 5 BIPL 38
-	1	5/.	0.00	to-and bacelt-	dans nobl	les & card		
	M.T.	Í		calebres (60% an	e mandia	2-5 40 4	Solit	S1001 37' Jo 40
5-				balk 1's > 80% be	self max	1'5 4/5 cm)	125	blows \$3650 rec
	37:40	2018	930	\$ 30-4090 M-VC	angelos :	sand(>75%	HEI	S # BIPL55
	3 43	1 201	34-20	6aself) # = 1070	VF-F Sa	nd t silt	Recon	en from 24' 39
240	2 24	1. 63 1.	888 A	150% felser no ix	n with He	1. Sl. merst.		+15-07 37-39
Report	ed By:	I. Horn		-96 -	Reviewed B	y: 1.0	Walks	F
		1.2	***		Title:			<
	Seda	arat .	11			Geologi:	T	
Signati	ure: V	11.	11 .	Date://-/07	Signature:	11/1/100		Date: 12/13/0;

	BOREHOLE	LOG			Page 2 of 4
ID: 15706	Mall Norse and				Date: //-/-07
	Well Name: 399-3-		Location: 300-FF.		
	TCE Investigation		Reference Measuring Point:	Ground	Surfice
Sample Gr			escription		Comments
Tune Diame	Log Group Name, Grai Color, Moisture Co Max Pa	in Size Dis ontent, So article Size	stribution, Soil Classification, rting, Angularity, Mineralogy, e, Reaction to HCI	Depth of C Method of Sample	Casing, Drilling Method, Driving Sampling Tool, er Size, Water Level
the see of 1	23.5-25.0:	Silt	1 Sender Sill m/an		mn 39,5'-42.5'
- X X 2 3	well- consoli	dated	H. vellowish brown	160 3	Aug 75%
- 4 33 24	80 (2.54, 6/4) si	Hwi	th irregular stringer	SHEIS	S# BIPL56
	therdened no	odules	of exidized	1-beal.	4.5, ~41-43'
- 335 2 -	o rellowish b.	rown	(10 YR, 5/6) silt.		
1 1 1 1 2	24 Silt 1	runs	Hons to SM with	DTWE	44.5 Las (11-5-07)
	200 The presen	ce of	V. th. sand.	c 11	
-	Dog No rxn wi	10 14	<u></u>	1. 11'	12001 42.7-45.7
	0:225.0'-355':	Ser	he lavauel 1-6	JETCH	BIP/57
	O. Same as a	navel	above silt laves	1-6-0	6.5. ~ 43-45'
	on mar #1		701	Archive	Serveles @ 40,45
	30			50,55.60	1.63,5.66,070,475
- W3 3T	35.5 - 47.0	: Sil4	4 Sundy Grovelland	+Capilla	us frinkae 115-07
	29 Poorly sorted	f une	ansolidated. Some		/ 0
	as gravel abo	ove, be	it with exchangive	Pamper	W.S. 48-50'
- N.J. 13	o weathing	£ 500	ondary clay. Alund	slough	48.5-50 bas
- V \$150	ant of brown	n (2.54	(3/5) (lay on subble	HETS	BIPLBZ
- < 444 .3	Date could sur	YALLS.		D	, ,
-	43.0 - 45.0 :	Sill	· La mal (1)	Pampet	W.5. 53,5-550
	20 -2' of with	11- 00	isolidated sitt/clay	THELS	* BIPLB3
175H.	250 gravel abou	e wat	in table silt	She La	Ana 53.5-55.0
	Cers light ut	Mowis	h brown (2.54, 6/3)	0	0
	-50% G 1	40%	1 -10%5	Solit S	000 55.5'-58.5
1	÷			346 610	405 60% rec.
2 11 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		Sandy	Gravel (1562)	HEIS	BIPL58
-1 2	Horly sorte	d f a	nconsolidated with		
12567. 0 1- 10070rt 1	Minor heavi	ng of	smell pobles	Pumped u	0.5. 62.5-64
	& couse s	A. 2	6070 sub- to	NO open	hole slough
7 1	la oto hater	1 de la	bles & small copples	HEIS #	2.0 to 64 Las
ine ine	- with < 40%	a mod	- V. coorse and.	Salit so	DON 62.5.65
Nº RGR	is besalt - Inn	SAN	A Trane Grand	Spiri Spi	175-11 1 Ham
	mel. sand	3 Al	ve- daminated. <17	1/2 line	as condened cill
	silt. NO rx	n we	the ACL.	HEIS	# NIPL59
	63.5-72.5:	Sandy	Sitt (SM)	1- Gal	not comple of
	Well-consolid	aled 1	4. d. grayth brown.	silt 62	5-65 (64-65)
	(2.5Y, 5/2.5) 62	the vi	25% silt \$ 125%	Split se	00n 65.3-67.8
	v. tine felsn	c ser		HEIS	BIPL60
ted By: J. Horne	r		Reviewed By: L.S.W	alker	
Grologist			Title: Geologist		
tura: 10 11.	Date: /		Simosture	10	Data: 12/2/
and the	and a state of the	T	Signature: AN Wal	1	Date: 12/13/07

				BOREHOLE LOG			Page <u>3</u> of <u>4</u> Date: (1-8-0-
Well ID	CS	706	N	ell Name: 399-:3-22	Location: 300 - FX	=-5 0	
Project	3	D-Are		, ,	Reference Measuring Poin		
	inc	Imple	-	1	Description	t: Groun	Comments
Depth	Туре	Blows	Graphic			- D - 11 - 1	
(Ft.)	No.	Recovery	Log		istribution, Soil Classification orting, Angularity, Mineralogy e, Reaction to HCI	, Depth of Method of Samp	Casing, Drilling Meth of Driving Sampling To pler Size, Water Level
80	~ 11		SF CON	-@ 69.5 sedi	ment is reduce	V . /	000n @ 81,5-84
	8666005 100% 100		с. 1	and is green	ish gray (Glay 7, 4/50	HEIS :	# 131PL61
-	1-19-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-			72.5 - 80.0: Silt	enish gray (Gley 2, 9		OA HEIS#BIR
15 _				\$ 1.0450/idaded	ith ~ 9090 folsic	-	· · · · · · · · ·
- 2				dominated (>907,	00 2 1	SILEINE	/
]		ad. s.T.		~1090 sitt. Sil	+ fraction gradual	the slough	85.6 - 86.0 Va
_				decieoses & san	ad grainsize	/	# BIPLBS \$
_		1		gradually increase	s with depth.		Duplicate - TBIP
10 -					/	Archiv	
_				80 -96 : Sano	(<u>(</u>)	@ 80,8	39,5,95,100,10
-		h.		Well-sonted g	on solidated (NE) #	109. #	112-119/695
-		ω.5.		greenish muy (6	1/2 2, 5/5GD) with	Slug +	est@ 77.5-8
-				100% ang fills	IC SAND (90% felser	seran	folled in to 7
15-				Sand is medium	increasing to	prior	to slug futing
-		1	00.00	mya romse a q.	5 693.	- /	0 0
-				96'-115':	S. J. I	Pumped	W.S. 93.5 - 45
-		:	0 0 D	Poorly sorted and	th 7759 hall	1 1 1	(93.5 - 94.0)
0_		0	000	fominated webbles	S & cohbles ~ 201		14.0-95.5
			200	m-vc and sond	medium sand	MILLE S	H 151P6156
		1	\cap	13 Selsic - domina	fed r-vi cand	Switter	d la bard bal
_		:(13 matic-dirminat	ed with = 590 M	. asel	B 1172-07 96 645
_		Č	2000	- minor heavi	A	in su	2 . IB Ngs
5_	*	3.5	0	- minor leaving	& from 106'- 107'	Pumpea	W.S. 105-112
_	Ē		0.0	0		w/ scree	n 105-111' 66
-	E	1 10	00			\$ slour	h 111'- 112'ba
-	E					HEIS	# BIPL 137
-	F		101				
0-	ľ.	10	000				
-	19	2	2060	7			
-			500		1 1 2		
-		0.0		115'-120': San	d (S)		
			0.04	Lell-sorda wear	uy consolidated		
5			-	with 29070 very	6 gray (Gley 1, 3/1)		
-		and the second se	4	1	live felsic sand		
-			1	~ 9970 felsic) 1	. th = 1070 silt.		
-	3	う際					
eported	By: J	Hor	ner	F	Reviewed By: L.D.	Valker	
tle: (eolo				Title: Geologis		
		1, mai			90010415		

			1.000	BOF	REHOLE LOG				Page <u>4</u> of <u>4</u>
Well ID): / 2	706		Well Name:	399-3-22	Location:	2000		Date: //-/3-07
Project				Investi	10000		300- EE		
	Sa	mple		6		Description	vieasuring Foint.	Ground	<u>Surface</u> Comments
Depth (Ft.)	Туре	Blows Recovery	Graph Log	ic Group I Color, N	Name, Grain Size Moisture Content, S Max Particle Si	(B)	il Classification, rity, Mineralogy,	Depth of (Method of	Casing, Drilling Method, Driving Sampling Tool, er Size, Water Level
120	с	8		36		C I I		Pumper	er Size, Water Level
		0	000	6: 120 -1" S. Por 4	sarled \$ 10	Sandy 19	wavel (msly	Screen.	ed 118-124.5 w/
	5	8		60-	75% an	ander to	sub-rad.	Stenger	164.5-125 665
-		8	000	a pento	les & copt	es (besal	+- demindel		
125-	8	1	ð	~ 607	log salt)	5 70 3	070 m-vc	Pumped	W.S. 128-135
	92		0.00	20 40 7	EP 11-15-07	0-15%	very dark	Screene	$\frac{(128-134.5)}{(34.5)-132}$
	4	w.3.	000	o gray	silt (Gele	41, 3/N	Narxn	- United	135 40
		$\langle \rangle$	9.0	y wit	n Hel.			Arching	e grab samples
130		2		20 	and the second s			coller Le	@ 125, 135,
		2	0				* **	1 140	log s
		1	0°C	9	· · · · · ·				
-		2	9	8 				an lat	
135-		19 <u>1</u>	00.4	- 135 - 1	1405 · C ·	+ / 44	2		
				- Well-	sould in	11-6040	RLM		
	•		~~~	muss	ive silt.	Dork and	enish no		
_				Coley 1	5/54) 42	ith mea	ium		
140			~~~	- p/115	tring. The	ue of	v. fine		
				is c	the mit	Hund 9	sol 41 F		
		Ϋ́,		ofc	ay.	inge !	CALLAR Z		
10 <u>-</u>					/				
_					-10 1				
-				-	Total dept	h = 140.5	bas		
	Ĩ								
_		1					22 		
						1.0			
_					·····				
_									
-									·
-						· · · · · ·			
		4							
Reporte		5. 1+or	her			Reviewed By	r. L.b.	Walke	*
	/	yais+	-		1000 PLAN - 100	Title:	Geologist	<u> </u>	
Signatur	re: 4	the p	An	<u> </u>	Date://-/5-0	Signature:	20 W	all	Date: 12/13/07
	\mathcal{O}								A-6003-642 (03/03)

				BOREHOLE LOG		Page of
Vell ID			144		Landian 200 FF	Date: 10-8-07
		5707		all Name: 399-4-14	Location: 300-FF-S	
roject		2 Ares	TCA		Reference Measuring Point:	Ground surface
Depth	Sa	mple	Graphic		Description	Comments
(Ft.)	Type No.	Blows Recovery	Log	Group Name, Grain Size D Color, Moisture Content, So Max Particle Siz	istribution, Soil Classification, orting, Angularity, Mineralogy, e, Reaction to HCI	Depth of Casing, Drilling Method, Method of Driving Sampling Tool, Sampler Size, Water Level
0-			20 (S () 4)	0-0.1 ; Concrete	1 11	Cuble tool duilling
			00		, 0	with drive barrel
4			00	0.1'- 3.5' : Sand	y Gravel (54)	4 113/4" Casing
-			000	Mod. sont. & dork	grayish brown with	
-	e - 1			50-60% Sal-roun	ded probles & couble	(arab sumples (archives).
-	S			# 40-50% m-c a	29. Sund ("70% ivon	- collected @ 2, 4, 10
-				with Hel (Extrem	to be balt). No ree	1 15,25,27,30, £ 35 Was
-			· 400.		consolidated (2.54, 1/2	Sample Q. 10' 645
1	a i		10.10	<u>13_ 63 6 (m, un</u>	Lonson and Logi I-	HEZST DIPL39
,	M.T.		0,40	3.5'-7.5: San	d (s)	nees anest
			000	Boorty sorted oliv	ue brown (2.5%, 4/4	M.T. QIS #RIPL40
	5 8		2000	sand (10070). San	dis of we with	
_	e 3		Stas	> 80% iron-stain	ed felsixs -2070	M.T. @ 20 # BIPL 41
_			1999	basalt. No exe	with Hel.	
-	M.T.		8000	.5' some small	pebble present	
-			000	- 7' irregular, st	ringens of very	Likely a paleosal @ the
-		6	500	_ pale brown 10	VR 7/3) Cally	dop of nave deprest
-	ų. 		Og.	- clay mainly	present on jebb	*
-	A.T.	+	09	Sertalles Var	strong ran up H	eg
- 1	A.1.	रे	0.00	It brown cliny	s one also present	7 4
-	ñ 9	1	03	The sandy	traver (SM)	M.T. @ 25 BIPL 4Z
-	ų.		0	colling softa an	Laubles estimate	
-		1 1	200	to sub and have	boulding, sub-roma	
	M.T.	न्द्र		-20% an	hereft dam san	(>30% conselv. course)
5-			60%	ssto silt over	it on reph le / entre	a construction of the cons
	M.T.	5	000	surfaces Color	rich class month	ong
		1	000	above continued	into arove to - 8,5	M.T. Q. 27 # BIP 143
_	Gral.	10% V	000	-15' silt fraction	on up to 5-1090	split spoon 28-31 bys
0-	1-02	. 4	0.0	fine send	also increases, stil	M.T. @ 31'Las taken
_	M.T.		80	~ 80% grev	el.	from s.s. shoe.
_	1 8	2.0	0.0	-24 very mg	ist, almost wet	S.S. # BIPL62
-	184	22	08	with It.	brown silt had	M.T. # BIPL44
_	A.T.	14 1	803e	ing the	moisture.	1-Gal G.S. 28-31 cleanout.
5-	5	142	988	~27 - 410 : Silt	1 roy Grand fist	S & v. moist / alterent wet
	1	1.3	200	roorly sorred with	70-801 Gounde	15.5. from 31-39 695
-	au.	1	2000	prover & coboos	10-BO	M.T. @ 34 b # BIRC
-			888	159 11 hand	hand a 10-	1-Gel, G.S. \$1-34'bas.
oport	od Dun			13/2 /r. 9104) 0	Paulawad Pur (1	
		J. Ho	nev		Reviewed By: L. D.	
itle:	Geor	logist	-		Title: Geologi's	4
ignatu	ire:	La He		Date: 10-4-07	1	Date: 12-17-07
	1					A-6003-642 (03/03)



			BOREHOLE LOG			Page 3 of 4
				T		Date: 10-11-07
Well ID:			ell Name: 399-4-14	Location: 300-	F.F.5 0	DU
Project:	300 Anea -	TCE :	Tavestigation (07)	Reference Measuring	Point: Gra	und Surface
Death	Sample	Quality	Sample	Description		Comments
(Ft.)	Type Blows No. Recovery	Graphic Log	Group Name, Grain Size D Color, Moisture Content, S Max Particle Size	Distribution, Soil Classific corting, Angularity, Miner	cation, Depth alogy, Metho	of Casing, Drilling Method, d of Driving Sampling Tool, mpler Size, Water Level
	Xki		13'-25' Sun	d (S)		Pio-11-07
0 -	GI 14 4.2	1.4	well- sould & an	AL (2.54 .5/1)	174	10-11-07
1.	- 401 7 1	2. Chiling + 1	~98% and to	sab-and med-	come Sale	+ 5000n 785'-81'
	BO Hand		felse sand (yes	To felsic 420	290. Look	s like sand from
1	00% red		metics with vi	in little base	et) abor	R 78.5 heaved in
85 -	2	0.0	Send is very sil	mila to ma	teix befo	ce sampling. 1804/pu
-		0.00	send from gra	velabove, bot	1007	a could all be
4	1	00	have abundant	MILAS (25%).~	290 5/04	sh, see FAR notes
_	1	000	t- in heruolithi	c pr bbles.	for	10-12-07.
_		CO0	75'-785' San	dy Grownel (sto) HE	IS#BIPL66
10		28	same as so	hove uno	alayer. 1-	Granon bucket
-	23	0.00	Had some trou	by with sum	a samp	163 @ 80-81 6gs
+	N		heaving while a	rilling gravel	Could and	81-12 1-12
-1.			naw uper mar	ix separating	from Pary	0
- +	24 715	0.00	gina ().			83,5-85.5 645
15	e	00	78.5 -85.5: 5	and (1007 8	(5)	
		000	V. well sorted	consolidated	dark Pum	ord w.s. 91-945 45
		000	arechist anone 16	s/ey 1. 4/1064) w	ith scree	n to 92' bas slough
_		00	121070 angular ve	-f felsic sur	1 92'-	94.5' bas. HETS #
100		20	f9870 felbre un	th <1070 sil	E 13/1	PLCS.
-		000	- 80' grein size	incleases to	m-c. A Su	itched to hard
_		000	Beveral len	ses of f.vf	sund tool	5 @ 105 by 5 10-1
-		00	still prese	nt	1 Tum	eed w.s. 104.5-1154,
-	13.5. ST.	20	85.5-91. 50	nay Gravel (SA EXPOS	ed server 104.5-111.7
105		00	mind- sorred with	- SOG- 7570	sour sloug	111.7 to 115 435.
4		00	Small cables	= 2.54 - 502 d	C. M. S	S-DIFLCCo
-		000	and Laker san	1/2007 AP	M-C SA	me intu
		000	> Goz Leleve t	dark areenish a	20 20	
110 -		0000	Gley 1, 4/5G).	Thin an u	41 And	ive samples e
		50	lunses present	+Q - 86' 60	1 .	79' 80', 85', 89'
	A A	08	90' has	0	91-97	93,100,105,110
_		00	91- 94 : Sand	(3) 10070	10-10-07 \$ 115	695
_	3 2	000	V. well-sorted	. angular m		
115_		30	v. durk greentst	gray felsed	8570	
-		500	sand.			
-	1	000	- 92'- drastic	color change y	rom	
-		Qet	Green to	in greensh	gray	
Reports	d By:	1.12.12. ····	Carey, of	Bowiewood Bur	1 112 11	
	d By: J. How	¥.×			. S. Walk	cr
Title: G	reologist			Titie: Geol	ogist	
	re: Ally Ho		Date 10-22-0	Signature:	2 mal	2 Date: 12/17/07

				BOREF	IOLE LOG				Page 4 of	
Vell ID:	re	707	w	ell Name: 39	7-4-14	Location: 3	00-FF			- 22-07
						Reference Mea	and the second second		ou	0
10,000		nple	ICE I	nvestigati		Description	sunny Font.	Groun		acl_
Depth	Туре	Blows	Graphic	Group Nam			location	Dooth of	Comments	_
(Ft.)	No.	Recovery	0002.00		ture Content, S Max Particle Si	Distribution, Soil Conting, Angularity ze, Reaction to Hereit	, Mineralogy,	Method of Same	Casing, Drillin of Driving Sam oler Size, Wate	pling Tool, <u>pling Tool</u> , <u>er Level</u>
20	Ë	A	-0	94'-129	S: Silty	Sandy (travel (mile	Hard y	col dri	ling
-	ł		000	Poorly .	3/1)	very dark	giry -	-	ble tool)	
		1	0000	i Grey	505 709	E consoli	zana	Fumpe	<u>w.s.</u>	20-131.5
1	į į	2	000	vounded	MALY	- down ne	hbles &	Expassed	30'44	claus l
15-			000	1 obbles	(7509	becalt)	30-40%	from	30' 40 1	JIS'EL
	ļ	1	zO	M-C	angular	felsia - de	m. sand	HETS	# BIPL	C. F.F.
1	Ē	1	e Se	(50-75	98 Selen) + 15-20	270 silt.			
4	F	1	0000	Nor	an with	n ACI		Slug	test on	10-23-0
_	F	1	0000					with	same s	etup
30-				129.5'-	136 : :	Siltfeloy ()	elm),	A.S 10.	5. deseri	bed at
_				Well-	sorted	multive	silt.			
-	3	_		Dark q	reentsh q	ray (Gley	1,5/56	Archive	samples	Q. 120'
-				Med i um	plast	city Tra	y of	125,1	30', F1	35 405
-	1			V. fine	Sand	is present		<u> </u>		
5-				-Mare:	Sample	taken trop	of clay			
-				ア	on hark	NOOL BIT				
-						- legz		·		
٦			1 1					1		
40			1							_
10-	1									
]	- 1									
_	1				-					
1	- 1		1 1					1		
45-	- 1					· · · · · · · · · · · · · · · · · · ·				
-	1		1 1					I		
4										
4										
4										-
50	1									
~										
-										
-										
55	1		l t					1		
-										
7										
]										
	4 0	/						(1) 1.		
		Hor	ner	· · ····		Reviewed By:	L.B.	Walk	re r	
	1000	1'ST	1			Title: 6	eologist		— <u> </u>	
lignatur	'e: 2	la the	·		Date: 10-24-0	Z Signature	as 11/0	. le	Date:	12/17/07

Distribution

No. of Copies

ONSITE

5 Fluor Hanford, Inc.

J. V. Borghese	E6-44
J. W. Lindberg	E6-35
S. P. Luttrell	E6-35
L. C. Swanson	E6-35
B. A. Williams	R3-60
CH2M HILL	

M. Connelly

H6-03

No. of

Copies

12 Pacific Northwest National Laboratory

J. S. Fruchter	K6-96
D. C. Lanigan	K6-75
D. R. Newcomer	K6-96
M. L. Rockhold	K9-36
R. M. Smith	K6-96
F. A. Spane	K6-96
P. D. Thorne	K6-96
V. R. Vermeul	K6-96
M. D. Williams	K6-96
S. B. Yabusaki	K9-36
Hanford Technical Library	P8-55